African Journal of **Business Management**

Volume 9 Number 18 28 September, 2015 **ISSN 1993-8233**







ABOUT AJBM

The **African Journal of Business Management (AJBM)** is published twice monthly (one volume per year) by Academic Journals.

African Journal of Business Management (AJBM) is an open access journal that publishes research analysis and inquiry into issues of importance to the business community. Articles in AJBM examine emerging trends and concerns in the areas of general management, business law, public responsibility and ethics, marketing theory and applications, business finance and investment, general business research, business and economics education, production/operations management, organizational behaviour and theory, strategic management policy, social issues and public policy, management organization, statistics and econometrics, personnel and industrial relations, technology and innovation, case studies, and management information systems. The goal of AJBM is to broaden the knowledge of business professionals and academicians by promoting free access and providing valuable insight to business-related information, research and ideas. AJBM is a weekly publication and all articles are peer-reviewed.

Contact Us

Editorial Office:	ajbm@academicjournals.org
Help Desk:	helpdesk@academicjournals.org
Website:	http://www.academicjournals.org/journal/AJBM
Submit manuscript online	http://ms.academicjournals.me/

Editor-in-Chief

Prof. Wilfred Isioma Ukpere

Department of Industrial Psychology and People Management, Faculty of Management, University of Johannesburg, South Africa.

Editors

Dr. Amran Awang Faculty of Business Management, 02600 Arau, Perlis, Malaysia

Prof. Giurca Vasilescu Laura University of Craiova, Romania 13, A.I. Cuza, 200585, Craiova, Dolj, Romania.

Associate Editors

Dr. Ilse Botha University of Johannesburg APK Campus PO Box 524 Aucklandpark 2006 South Africa.

Dr. Howard Qi *Michigan Technological University 1400 Townsend Dr., Houghton, MI 49931, U.S.A.*

Dr. Aktham AlMaghaireh United Arab Emirates University Department of Economics & Finance United Arab Emirates.

Dr. Haretsebe Manwa University of Botswana Faculty of Business University of Botswana P.O. Box UB 70478 Gaborone Botswana.

Dr. Reza Gharoie Ahangar Islamic Azad University of Babol, Iran. **Dr. Sérgio Dominique Ferreira** *Polytechnic Institute of Cavado and Ave Campus IPCA, Lugar does Aldão, 4750-810. Vila Frescainha, Portugal.*

Prof. Ravinder Rena Department of Economics University of the Western Cape Private Bag: X17 Modderdam Road Bellville 7535 Cape town, South Africa

Dr. Shun-Chung Lee Taiwan Institute of Economic Research No. 16-8, Dehuei Street, Jhongshan District, Taipei City 104, Taiwan.

Dr. Kuo-Chung Chu National Taipei University of Nursing and Health Sciences No. 365, Min-Te Road, Taipei, Taiwan.

Dr. Gregory J. Davids University of the Western Cape Private Bag x17, Bellville 7535, South Africa.

Prof. Victor Dragotă Bucharest Academy of Economic Studies, Department of Finance Bucharest, Sector 1, Piata Romana no. 6, Room 1104, Romania

Dr. Maurice Oscar Dassah School of Management, IT and Governance University of KwaZulu-Natal Post Office Box X54001 Durban 4000 South Africa. **Prof. Joseph Offiong Udoayang** University of Calabar P.M.B 1115, Calabar. Cross River State, Nigeria.

Prof. Robert Taylor University of KwaZulu-Natal Varsity Drive, Westville South Africa.

Dr. Nazim Taskin Massey University - Albany Quad Building A, Room 3.07 Gate 1, Dairy Flat Highway (State Highway 17)Albany, New Zealand

Prof. João J. M. Ferreira University of Beira Interior (UBI) Estrada do Sineiro, Pólo IV 6200 Covilhã, Portugal.

Dr. Izah Mohd Tahir Universiti Sultan Zainal Abidin Gong Badak Campus, 21300 Kuala Terengganu, Terengganu, Malaysia.

Dr. V. Mahalakshmi Panimalar Engineering College 7-A,CID Quarters, Mandaveli,Chennai-600028, Tamilnadu, India.

Dr. Ata Allah Taleizadeh Iran University of Science and Technology Faculty of Industrial Engineering, Iran University of Science and Technology, Narmak, Tehran, Iran.

Dr. P.S. Vohra Chandigarh Group of Colleges, Landran, Mohali, India #3075, Sector 40 D Chandigarh, Pin code 160036

Dr. José M. Merigó University of Barcelona Department of Business Administration, Av. Diagonal 690, Spain.

Prof. Mornay Roberts-Lombard Department of Marketing Management, C-Ring 607, Kingsway campus, University of Johannesburg, Auckland Park, Johannesburg, 2006, South Africa **Dr. Anton Sorin Gabriel** Carol I Boulevard, No. 11, 700506, Iasi, Alexandru Ioan Cuza University Iaşi, Romania.

Dr. Aura Emanuela Domil 31 Horia Creanga, zip code 300253, Timisoara, West University from Timisoara, Faculty of Economics and Business Administration, Romania.

Dr. Guowei Hua NO. 3 Shangyuancun, Haidian District, Beijing 100044, School of Economics and Management, Beijing Jiaotong University, China.

Dr. Mehdi Toloo Technical University of Ostrava, Ostrava, Czech Republic

Dr. Surendar Singh Department of Management Studies, Invertis University Invertis village, Bareilly -Lucknow Highway, N.H.-24, Bareilly (U.P.) 243 123 India.

Dr. Nebojsa Pavlovic High school "Djura Jaksic" Trska bb, 34210 Raca, Serbia.

Dr. Colin J. Butler University of Greenwich Business School, University of Greenwich, Greenwich, SE10 9LS, London, UK.

Prof. Dev Tewari School of Economics and Finance Westville Campus University of Kwa-Zulu Natal (UKZN) Durban, 4001 South Africa.

Dr. Paloma Bernal Turnes Universidad Rey Juan Carlos Dpto. Economía de la Empresa Pº de los Artilleros s/n Edif. Departamental, Desp. 2101 28032 Madrid, España

Dr. Jurandir Peinado Universidade Positivo Rua Silveira Peixoto, 306 Zip 80240-120 Curitiba – PR – Brazil **Prof. Fabrizio Rossi** University of Cassino and Southern Lazio (Italy) Via G. Di Biasio 43, Cassino (Italy)

Editorial Team

Dr. T.S. Devaraja Department of Commerce, Post Graduate Centre, Hemagangotri Campus, University of Mysore India.

Dr. Peide Liu Business Administration School, Shandong Economic University, China

Dr. Marwan Mustafa Shammot King Saud University, P.O.Box 28095, Riyadh 11437 Kingdom of Saudi Arabia.

Dr. Hela Miniaoui University of Wollongong in Dubai, Knowledge Village, Block 15 PoBox 20183, Dubai UAE

Dr. Suhanya Aravamudhan 6965 Cumberland Gap Pkwy, Harrogate, TN USA

Dr. Hooman Attar Amirkabir University of Technology Iran

Prof. Luis Antonio Fonseca Mendes University of Beira Interior – Business and Economics Department -Estrada do Sineiro – Polo IV – 6200-209 Covilhã Portugal

Wu, Hung-Yi Department of Business Administration Graduate Institute of Business Administration National Chiayi University No.580, Xinmin Rd., Chiayi City 60054, Taiwan (R.O.C.)

Shu-Fang Luo No.28, Da-Ye S. Road, Lin-Hai Industrial Park, Hsiao-Kang, 812, Kaohsiung City Taiwan Ahmad.M.A.Ahmad Zamil King Saud University, P.O.Box 28095 , Riyadh 11437 Kingdom of Saudi Arabia

Olof Wahlberg Mid Sweden University, 851 70 Sundsvall Sweden

Mario Javier Donate-Manzanares Facultad de Derecho y Ciencias Sociales Ronda de Toledo, s/n 13071 Ciudad Real Spain

Mohamed Abd El Naby Mohamed Sallam Faculty of Commerce -University of Kafr El-Sheikh Egypt

Dr. Bhaskar Bagchi Alipurduar College -Helapukur (Shibmandir); CHANDERNAGAR, Pin – 712136; West Bengal INDIA

Dr. Pawel Tadeusz Kazibudzki Jan Dlugosz University in Czestochowa, The Faculty of Social Sciences Poland

Dr. Cherukuri Jayasankaraprasad Department of Business Management Krishna University (State Govt. of A.P.) Machilipatnam, A.P., India-521001

Dr. Aleksander Aristovnik Faculty of Administration, University of Ljubljana, Slovenia

Dr. Bhavesh Parmar Department of Business Management, Sankalchand Patel College of Engineering, Visnagar. (Affiliated to Gujarat Technological University.) India.

Prof. Paulo Alves Lisbon College of Accountancy and Administration and Lusofona University Portugal

Dr. Mathew Analogbei The Open University Business School UK. Centre for Marketing & Strategy, The Open University Business School, Walton Hall, Milton Keynes, MK7 6AA, United Kingdom



academic Journals

Vol. 9(18), pp. 633-640, 28 September, 2015 DOI: 10.5897/AJBM2015.7874 Article Number: 11BE16155246 ISSN 1993-8233 Copyright © 2015 Author(s) retain the copyright of this article http://www.academicjournals.org/AJBM

African Journal of Business Management

Full Length Research Paper

Financial development and economic growth in Togo

FANTESSI Amavi Agbelenko

Universite de Kara-Togo.

Received 11 July, 2015; Accepted 18 August, 2015

This study investigated the link between financial development and economic growth in Togo for the period 1981-2010. The study employed Johansen's cointegration and Granger causality testing procedure in the context of Vector Error Correction Method (VECM). The result shows a positively and statistically significant effect of financial development on economic growth in Togo and Granger causality test supports supply-leading view that comes from financial development to economic growth. This means that the causal relationship between financial development and economic growth in Togo is unique and is running from financial development to economic growth. The empirical results further confirm a unique cointegration relationship among real GDP per capita, financial development, inflation, primary completion rate, openness, foreign direct investment and real exchange rate. In addition, the variable primary completion rate, foreign direct investment and real exchange rate contribute positively to economic growth in Togo while inflation and openness discourage the economic growth in the Togo. The findings of the study call for the introduction of effective policy measures to deepen financial sector since financial development is an essential precondition for the positive impact of FDI on economic growth.

Key words: Financial development, economic growth, devaluation, Togo.

INTRODUCTION

A significant number of empirical studies have attempted to investigate the link between financial sector development and the economic growth. The neoclassical theory early raised the impact of savings on economic growth through capital accumulation (Solow, 1956; Stiglit and Hirofumi, 1969). The endogenous growth theory reignites the debate between economic growth and financial sector development. Then a growing body of literature attempts to highlight the direction of the relationship between financial development and economic growth. Some literature argue that, economic growth just follow finance development through the supply of financial service that come from growth of financial institutions and markets (McKinnon-Shaw, 1973;

E-mail: fantessi@yahoo.fr. Tel: (+228)92708147/90241424.

Author(s) agree that this article remain permanently open access under the terms of the <u>Creative Commons</u> <u>Attribution License 4.0 International License</u> King and Levine, 1993b; Laima and Oleksandra, 2014). On the contrary, some people argue that the growth of modern financial institutions is a result of the demand of financial services that come from investors and savers (Robinson, 1952; Sunde, 2013). According to Patrick (1966), Nabila and Zakir (2014), the relationship is bidirectional causality between financial deepening and economic growth. While, some other authors argue that there is no link between financial development and economic growth (Lucas, 1988; Mohamed, 2008). However, this argument seems not to be true since when we experience financial crisis, countries economic growth tend to slow down. This was evident in the recent USA financial crisis which was contagious to other countries through globalization and economic integration. Though a large amount of work that investigated the link between financial development and economic growth; there is no consensus on the impact of financial development on economic growth and few papers have examined the relation between financial development and economic growth in Togo. Furthermore, the recent financial crisis has shown that the impact of the financial development on economic growth depends on the level of economic development of a particular country because countries have different level of economic development and cannot use the same macroeconomic policy to fight against the crisis. Hence, it is crucial that each country understands how its financial system relates to its economic growth in view to find the right macroeconomic policy that can assist during the financial crisis. In spite of the prominent role of the financial sector in influencing economic growth, Togo is dominated by banks and still characterized by regional underdeveloped financial markets (BRVM) that constrains resource mobilization and hinder economic growth. Besides, there were privatizations of financial institutions in recent past, which were initially owned by government. It is for these reasons that this study is set out to investigate the impact of financial development on economic growth in Togo.

This paper departs from previous studies by utilising new data since Rousseau, (2007) argued that financegrowth is not as strong in more recent data as it was in the original studies with data for the period from 1960 to 1989. The study further employs modern econometric procedures that address for potential biases induced by simultaneity, omitted relevant variables, non stationary in data and cointegration issue that have been persistently featured in the previous study. Besides, Togo has experienced a severe political crisis from 1990 to 2005 while country's stability is utmost important for financial development. In addition, this study used one variable as a proxy of financial development; namely domestic credit to private sector as a percentage of GDP. The motivation of using domestic credit to private sector as a proxy for financial development is justified as follow. First, domestic credit to private sector as a percentage of GDP refers to financial resources provided to the private sector and credit measures the extent to which financial market and institutions provided credit to private sector activities. Second, Togo is under fixed exchange rate, while fixed exchange rate means giving up monetary policy. The last justification is due to the neutrality of money in the long run since the long run economic growth is due to technical progress; capital accumulation and skill labor (neoclassical and endogenous growth theory).

Moreover, a panel or cross country study on the subject may not highlight the specificities of each country that may influence the relationship since it estimates only average effects for a sample of countries covered. Therefore, this study is set out to explore the relationship between financial development and economic growth in Togo while taking into account the endogeneity issue and the currency devaluation.

The rest of the paper is organized as follows. Section 2 reviews the existing literature. Section 3 presents the methodology that enables us to investigate the impact of financial development on economic growth in Togo. Section 4 describes data set, presents the estimation technique and interpretation of the results. Section 5 concludes the study and provides some policies recommendations.

LITERATURE REVIEW

Dimitris et al. (2003) investigated the relationship between financial development and economic growth. They used data of 10 developing countries over the period 1970 to 2000. They have used the ratio of total bank deposits liabilities to nominal GDP as measure of financial development. Their findings showed that long run causality runs from financial development to economic growth. They further argued that there is no short run causality between financial development and output.

Christopoulos and Tsionas (2004) used data of 10 developing countries to determine the relationship between financial development and economic growth. The authors used panel data analysis and the ratio of total bank deposits liabilities to nominal GDP as a proxy of financial development. They found evidence of longrun causality running from financial depth to economic growth.

Valenzuela (2009) examined the Impact of Financial Development on Economic Growth for Chile. The author argued that financial development has a significant effect on GDP and the most important channel is through total factor productivity.

Sanusi and Salleh (2007) determined the relationship between financial development and economic growth in

Malaysia, using data from 1960 to 2002. Ratio of broad money to GDP, credit provided by the banking system and deposit money banks to GDP were used as proxy of financial development. The autoregressive distributed lag model was used in their analysis. The result showed that the ratio of broad money to GDP and credit provided by the banking system exert positive and statistically influence on economic growth in the long-run.

Laima and Oleksandra (2014) analysed the relationship between financial development and economic growth in Lithuania. The authors concluded that the relationship between financial development and economic growth in Lithuania is consistent with the supply leading hypothesis meaning that there is the significant impact of Lithuanian financial development on country's economic growth.

Fatima (2004) examined the link between financial development and economic growth in Morocco over the period 1970 to 2000. The author used three financial indicators as measure of financial development, namely ratio of liquid liabilities (M3) to GDP, ratio of domestic credit provided by banking sector to GDP and domestic credit to private sector to GDP. The result from Granger-causality showed a short-run relationship between financial deepening and economic growth.

Khan and Shenhadji (2003) studied the relationship between financial development and economic growth by using data of 159 countries. The authors have used cross –section data framework and the study covered period 1960 to 1999. They further used two-stage least square (2SLS) to solve endogeneity problem in the underlying relationship. The result indicated a positive and statistically significant impact of financial development on economic growth.

Esso (2009) examined the causal relationship between financial development and economic growth in ECOWAS countries, using data from 1960 to 2005. The author adopted the autoregressive distributed lag analysis and the test for non causality proposed by Toda and Yamamoto (1995) were used. The ratio of M2 to GDP was used as measure of financial development and the author found long-term relationship between financial depth and economic growth in some countries namely, Cote d'Ivoire, Guinea, Niger and Togo. But a negative long-run relationship was found in Sierra Leone and Cape Verde.

Nabila and Zakir (2014) studied Financial Development, Trade Openness and Economic Growth in Developing Countries. The result showed that there are strong evidences of the long-run relationship between financial development and economic growth in developing countries. In addition, the study revealed a bi-directional causation between financial development and FDI. Furthermore, trade openness has impact on financial development in all the countries, which calls for the introduction of effective policy measures to promote trade between countries.

Christophe et al. (2008) investigated the finance-growth relationship. They used a sample of 63 industrial and developing countries over the period 1960 to 1995 and 1960 to 2000. They have used three indicators as measure for financial development namely, private credit by deposit money banks to GDP; liquid liabilities of the financial system to GDP and the credit by money banks and other financial institutions to the private sector. Their findings support the one way causality from economic growth to financial development.

Zang and Kim (2007) examined whether financial development precede economic growth. The authors used data of seventy four countries over the period 1961-1995. They used three financial indicators as measure of financial development, namely ratio of liquid liabilities to GDP, ratio of deposit money bank domestic assets to deposit money banks domestic assets plus central bank domestic assets and credit issued to private enterprises as a share of GDP. Their findings confirm that economic growth precedes subsequent financial development.

Luintel and Khan (1999) examined finance-growth link for a sample of 10 less developed countries. The author used multivariate vector autoregression analysis to address the relationship between financial depth and economic growth. Their findings showed two ways causality between financial depth and economic growth.

Abu-Bader and Abu-Qarn (2006) investigated the causal relationship between financial development and economic growth in five (5) Middle Eastern and North Africa (MENA) countries for different period ranging from 1960 to 2004. Their findings gave a weak support for the supply-leading view. But the Granger causality tests indicated evidence of either bidirectional or one way running from economic growth to financial development.

Fang and Jiang (2014) studied the Promoting Effect of Financial Development on Economic Growth in China. Empirical results showed that the banking and insurance sectors provide significant promoting effects on economic growth; the promoting effect of the securities sector is uncertain.

Zang et al. (2010) have studied the finance-growth nexus for South Korea by employing error correction model and a nonlinear smooth transition error correction technique. The authors have used a quarterly data from 1970 to 2004. The cointegration test shows that the financial development has a positive impact on the longrun economic growth and a causal relationship is bidirectional in the long run. While their empirical results do not reveal any positive influence in the short run.

Houssem and Hassene (2011) addressed the Causality between Financial Development and Economic Growth. The authors have used Fully modified OLS approach, GMM analysis and data of 10 countries over 1990-2006, six countries from OECD and four others from MENA. A long-run relationship between financial development and economic growth was confirmed for the 10 countries investigated by a panel data cointegration analysis. The GMM system approach showed a strong positive link between financial development and GDP per capita. The bi-direction causality between financial development and economic growth was found for OECD countries when error correction model approach was used, while a unidirectional causality was found for MENA countries that run from economic growth to financial development.

Adamopoulos (2013) analysed financial development and economic growth and has used a vector error correction model, based on Johansen cointegration analysis and stationary tests. Finally, Granger causality method was applied in order to define the direction of causality between the examined variables. The empirical results indicated that there is a bilateral causal relationship between economic growth and industrial production, while there is a unidirectional causality between economic growth and credit market development. Also, stock market development causes economic growth and industrial production.

Rousseau and Wachtel (2007) showed that the finance-growth relationship is not strong in more recent data as it was in the original studies with data for the period over 1960-1989. They argued by saying; firstly, excessive financial development or too rapid growth of credit may lead to both inflation and weakened banking systems which in turn inhibit growth through financial crisis (case of USA subprime from summer 2007). Secondly, excessive financial liberalization from 1980 and 1990 in countries that lacked the legal or regulatory infrastructure to exploit the benefit successfully.

Mohamed (2008) in his study of finance-growth nexus in Sudan from 1970 to 2004, has used the autoregressive distributed lag framework. He used two measures of financial development, namely the ratio of M3 to GDP and the ratio of credit to the private sector to GDP. The study showed a weak relationship between financial development and economic growth.

Ernesto and Marcelo (2013) analysed Financial Development and Economic Growth for a sample of 111 countries covering the period 1961 to 2009. The authors used time series techniques that deal with the problem of cross-sectional dependence and relax the constraint that the slope parameters of the independent variables should be the same for all countries. They demonstrated that financial development does not have a significant influence upon economic growth.

Keho (2010) examined the Effect of Financial Development on Economic Growth for seven out of eight countries of UEMOA. The author found no evidence of nonlinear relationship between finance and growth. The author used threshold models, cross-terms regression and argued that financial development has no significant impact on economic growth regardless of the level of inflation.

In light of the literature reviewed above, it is evident that the existing literature present mixed results in that, some report positive, negative while others report no relationship between financial development and economic growth. This implies therefore there is still need to further investigate the relationship between financial development and economic growth specifically for country level like Togo in order to provide policy maker a better guide to formulate policy (ies).

METHODOLOGY

In this section we present the method that enabled us to examine the relationship between financial development and economic growth in Togo for the period 1981 to 2010. The study employed two different models based on Johansen cointegration test. The first model examines the long run equilibrium relationship among the variables and the second investigate the short run equilibrium.

Theoretical assumptions

In order to achieve the objective of the study, Financial Development, Primary completion rate, Devaluation and Foreign direct investment are expected to have positive effect on economic growth in Togo while Inflation and Openness are expected to have a negative effect on economic growth in Togo.

Model specification

The empirical model for the long run equilibrium:

$$\begin{aligned} & \text{Ly}_{t} = \beta_{0} + \beta_{1} \text{ Fd}_{t} + \beta_{2} \ln f_{t} + \beta_{3} \text{ Pm}_{t} + \beta_{4} \text{ Op}_{t} + \\ & \beta_{5} \text{ LFDI}_{t} + \beta_{6} \text{ Rexch}_{t} + \text{e}_{t} \end{aligned}$$

 Ly_t = natural logarithm of real GDP per capita,

 Fd_t = measure of financial development proxy by variable credit facilities available to domestic sector,

 Inf_t = inflation, consumer prices (as an annual percentage),

 Op_t = openness measured as exports plus imports as a percentage of GDP,

 Pm_t = primary completion rate used as measure of human capital development,

LFDI, = natural logarithm of foreign direct investment,

Rexch $_{t}$ = real exchange rate between USA dollars and franc CFA, variable introduced

to capture the effect of the franc-CFA devaluation,

 $e_t = error term.$

Model for the short run equilibrium

 γ resid $_{t-1}$ + \mathcal{U}_t

The short run model is specified as follows:

$$\Delta \operatorname{Ly}_{t} = \delta_{0} + \delta_{1} \Delta \operatorname{Fd}_{t} + \delta_{2} \Delta \operatorname{Inf}_{t} + \delta_{3} \Delta \operatorname{Pm}_{t} + \delta_{4} \Delta \operatorname{Op}_{t} + \delta_{5} \Delta \operatorname{LFDI}_{t} + \delta_{6} \Delta \operatorname{Rexch}_{t} +$$

Where:

resid_t = Ly_t -
$$\hat{Ly}_t$$

 $\hat{Ly}_t = \hat{B}_0 + \hat{B}_1 \operatorname{Fd}_t + \hat{B}_2 \operatorname{Inf}_t + \hat{B}_3 \operatorname{Op}_t + \hat{B}_4 \operatorname{Pm}_t + \hat{B}_$

 B_5 LFDI $_t + B_6$ Rexch $_t$

resid, is the residual from the long run estimation.

DATA, ESTIMATION TECHNIQUE AND INTERPRETATION OF THE RESULTS

Data

The data for this study (Financial development and Economic growth in Togo) have been collected from the World Development Indicators 2011 (World Bank), except of exchange rate between USA dollars and franc-CFA which has been collected from UNCTAD data base.

Estimation technique

Since we are using time series data for estimating the impact of financial development on economic growth in Togo, the estimation technique is very crucial. The major concern with the time series is that if non-stationary of data series persists then it may lead to spurious relationship. In order to avoid spurious regression, we have checked for unit roots (Appendix: Table 3) for the variables using Augmented dickey fuller (ADF) and we found that variable inflation, primary completion rate and real exchange rate are stationary but log real GDP per capita, financial development, openness and foreign direct investment are not stationary but are integrated of order one; so risk of cointegration which might lead to spurious regression. Then we addressed for cointegration issue by using Johansen (1988) procedure and the result revealed unique cointegration relationship among variables (Appendix: Table 4). Moreover, we confirmed the cointegration issue through the unit root test of the residual from the long run regression (Appendix: Table 5). This justifies the use of the vector correction model in our study. Furthermore we used Granger causality test to address the direction of causality between financial

Table 1	. Long-run	relationship.
---------	------------	---------------

Variables	Ly_t
	0.0216***
Fd_t	(0.00705)
IC	-0.0163***
Inf_t	(0.00316)
D	0.000535
Pm_t	(0.00141)
0	-0.00451***
Op_t	(0.000884)
	0.0172*
	(0.00963)
Poych	4.14e-06
Rexcin _t	(4.34e-06)
Constant	5.436***
Constant	(0.264)
Observations	30
R-squared	0.735

*** p<0.01, ** p<0.05, * p<0.1. Robust standard errors in parentheses. Note: Natural logarithm of real GDP per capita (Ly_t) is the dependent variable.

development and economic growth and the result confirms unique causality between financial development and economic growth in Togo which runs from financial development to economic growth (Appendix: Table 6). Besides, Ramsey Reset test for better specification of the model has showed a p-value of 86.97% higher than 10%. This means a better specification of the model and no omitted variable bias.

EMPIRICAL RESULT AND INTERPRETATION

Estimation of the long-run relationship

The estimation of the long-run relationship is presented in Table 1. The variable financial development (credit to domestic sector as percentage of GDP) is positively and statistically significant at one percentage (1%) level. This means financial system improvement is crucial for sustainable economic growth in Togo. This positive effect of financial development on economic growth is theoretically and empirically supported in the literature by McKinnon (1973) and Laima and Oleksandra (2014).

The variables primary completion rate and real exchange rate have the positive sign as expected but not significant, while foreign direct investment is positive and statistically significant at 10% level. The positive effect of primary completion rate has been supported by some empirical study which argued that a higher literacy rate improves the efficiency of an economy as it provides a more productive labour force (Wilson and Briscoe, 2004; Berthelemy et al., 1996). A possible explanation of the insignificance of primary completion rate is due to the proxy variable which does not capture fully human capital development. Possibly if we had a data for high school or university could serve as a good proxy for human capital development.

The variable real exchange rate, although insignificant contribute positively to economic growth in Togo. This means this country did not reap the full benefit of devaluation due to the uncompetitive products because of underdeveloped infrastructure. The positive effect of devaluation on economic growth has been support by Huseyin et al. (2008) and Hala and Sahar (2008).

The foreign direct investment is statistically significant at 10% level which means the flow of FDI is adequate to stimulate economic growth in Togo as the FDI is the way the technology was transferred to less developing countries and the aspect of employment generation.

The variable inflation and openness have negative sign as expected and are statistically significant at 1% level. This means inflation and openness are the main factors that deter economic growth in Togo. The negative impact of inflation on economic growth in Togo is when prices increase persistently, the cost of production will be expensive to the investors, hence can force them to reduce their investment plan. This implies they reduce production and employment and thus will have a negative effect on economic growth. This negative effect has been supported by Bruno and Easterly (1996), Gylfason (1999) and Rousseau et al. (2000).

A greater degree of openness may benefit for Togo if its opening economy will enhance competitions, thus improve the quality and stabilize price of goods and services and promotes the domestic export but this is not the case in the country under study. This country is exporter of raw materials, agriculture and intermediate goods and is importer of final goods due to underdeveloped industrial sector. This kind of trade will negatively affect the economic growth of the country, because it profit only for foreign companies which are supplier of final goods to grow in size by increasing their production.

Estimation of the short-run relationship

The results for the short run model presented in Table 2 shows that financial development, primary completion rate and foreign direct investment maintain the same sign, positive as in the long run while inflation and openness maintain the same sign, negative, just like in the long run results. Table 2. Short-run relationship.

Variables	ΔLy_t
	0.0271***
Δ Fd $_t$	(0.00942)
4 1-4	-0.0102***
$\Delta \inf_t$	(0.00302)
4 D	0.000171
ΔPm_t	(0.000647)
	-0.00329***
ΔOp_t	(0.000913)
	0.00432
	(0.00420)
A Dovoh	-2.11e-06
$\Delta \operatorname{Rexch}_{t}$	(2.88e-06)
lograpid	-0.508**
lagresiu	(0.182)
Constant	0.00246
Constant	(0.0228)
Observations	29
R-squared	0.790

*** p<0.01, ** p<0.05, * p<0.1. Robust standard errors in parentheses. Note: First difference of natural logarithm of real GDP per capita (ΔLy_r) is the dependent variable.

The error correction term has the expected sign, negative and statistically significant at 5% level. This implies that 50.8% of the deviation from the long run equilibrium is corrected each period (annually).

CONCLUSION AND RECOMMENDATIONS

The study examined the relationship between financial development and economic growth in Togo for the period 1981-2010. Togo time series investigation shows unique cointegration relationship among real GDP per capita, financial development, inflation, primary completion rate, openness, foreign direct investment and real exchange rate. In addition, the results confirm a positive impact of financial development on economic growth and the Granger-causality test give support to unidirectional causality that runs from financial development to economic growth.

Furthermore, the variable primary completion rate, foreign direct investment and devaluation of franc-CFA proxy by real exchange rate between USA dollars and franc-CFA, contributed positively to economic growth in Togo while inflation and openness discourage economic growth in Togo. The positive impact of education on economic growth has been support by some author among them; we have Berthelemy et al. (1996) who argued that human capital development is pre-condition for economic growth. The negative impact of inflation and openness on economic growth has been found by Lucas (1998); Rivera-Batiz and Xie (1993); Rousseau et al. (2000) and Gylfason (1999).

From policy perspective, government of Togo should adopt the following appropriate policy: It is therefore imperative that policies which seek to increase financial depth be vigorously pursued in order to stimulate economic growth. The government needs to introduce an effective reform in education sector to enhance human resources. Finally, the government needs to control for the opening of the country economy and pursue tight monetary policy so as to lower the inflation.

In summary, these findings highlight the relationship between financial development and economic growth in Togo and the main limitation of the study is it does not include a variable which may account for country risk level due to data problems.

Conflict of Interests

The author has not declared any conflicts of interest.

REFERENCES

- Abu-Bader S, Abu-Qarn A (2006). "Financial development and Economic growth nexus:Time series evidence from middle eastern and north African countries". Munich Personal RePEc Archive (MPRA) Paper 972, University Library of Munich, Germany.
- Adamopoulos A (2013). "Financial Development and Economic Growth: A Revised Empirical Study for Ireland"European Research Studies, pp. 25-33 Volume XVI, Issue (2): 2013.
- Christopoulos D, Tsionas E (2004). "Financial development and economic growth: Evidence from panel unit root and cointegration tests". J. Development Econ. 73: 55-74.
- Dimitris K, Efthymios G (2003). "Financial development and economic growth: evidence from panel unit root and cointegration tests". J. Development Econ. 73: 55-74.
- Esso L (2009). "Cointegration and causality between financial development and economic Growth: Evidence from ECOWAS countries". Eur. J. Econ. Financ. Administrative Sci. 16.
- Ernesto R, Marcelo P (2013). "Finance and economic growth: new evidence from time series analysis (1961-2009)" *Applied Economics Letters*, 20 (7-9): 893-896.
- Fang X, Jiang Y (2014) "The Promoting Effect of Financial Development on Economic Growth: Evidence from China", *Emerging Markets Finance & Trade / January*–February 50 (1): 34–50.
- Fatima A (2004). "Does financial development cause economic growth? An empirical Investigation drawing on the Moroccan experience". Working papers 000295, Lancaster University Management School, Economics Department.

- Houssem R, Hassene B (2011). "The causality between financial development and Economic growth: Panel data cointegration and GMM System Approaches". Int. J. Econ. Finance. 3: 1.
- Keho Y (2010). "Effect of financial development on economic growth: Does inflation matter? Time series evidence from the UEMOA Countries". Int. Econ. J. 24 (3): 343-355.
- Khan S, Shenhadji A (2003). "Financial development and economic growth: A review and new evidence". Journal of African Economies, 12, AERC Supplement 2: ii89-ii110.
- Laima U, Oleksandra S (2014). "The Relationship between Financial Development and Economic Growth in Lithuania", *Transformations in Business & Economics*. 13 (33C): 446-467.
- Luintel K, Khan M (1999). "A quantitative reassessment of the finance growth nexus: evidence from a multivariate VAR". J. Devel. Econ. 60(2), 381-405.
- Lucas R (1988). "On the mechanics of economic development ". J. Monetary Econ. 22(1): 3-42.
- Mohamed S (2008). ⁴Financial-growth nexus in Sudan: empirical assessment based on an autoregressive distributed lag (ARDL) model". Arab Planning Institute Working Paper series 0803.
- Nabila A, Zakir H (2014). "Financial Development, Trade Openness and Economic Growth in Developing Countries Recent Evidence from Panel Data" Pakistan Economic and Social Review Volume 52, No. 2 (Winter 2014), pp. 99-126.
- Patrick H (1966). "Financial development and economic growth in underdeveloped countries". Economic Development and Cultural Change, 14(1): 174-189.
- Sanusi N, Sallah N(2007). "Financial development and economic growth in Malaysia: An Application of ARDL approach", [Online], available:

http://www.ibacnet.org/bai2007/Proceedings/Papers/2007bai7443.do c.

Zang H, Kim Y (2007). "Does financial development precede growth? Robison and Lucas Might be right". Applied Economic Letters, 14: 15-19.

Variables	ADF Test	Integration order	1% CV	5% CV	10% CV
Ly	-4.323	l(1)	-3.736	-2.994	-2.628
Fd	-3.808	l(1)	-3.736	-2.994	-2.628
Inf	-3.513	l(0)	-3.73	-2.992	-2.626
Pm	-3.371	l(0)	-3.73	-2.992	-2.626
Ор	-3.83	l(1)	-3.736	-2.994	-2.628
FDI	-4.133	l(1)	-3.75	-3	-2.63
Rexch	-4.746	l(0)	-3.73	-2.992	-2.626

Table 3. Augmented Dickey-Fuller test.

Table 4. The results of Johansen cointegration test.

Trend: constant			Number of o	obs = 29	
Sample:	1982-20	10			lags=1
Max rank	Parms	LL	Eigenvalue	Trace statistic	5% CV
0	7	-798.49994		130.6383	124.24
1	20	-778.35102	0.75082	90.3405*	94.15
2	31	-762.39597	0.66724	58.4304	68.52
3	40	-751.36143	0.53280	36.3613	47.21
4	47	-743.08274	0.43501	19.8039	29.68
5	52	-737.12599	0.33689	7.8904	15.41
6	55	-734.09165	0.18882	1.8217	3.76
7	56	-733.18079	0.06089		

 $\label{eq:table 5. Unit root test for the residual from the long run estimation.$

Variable	ADF Test	Phillips Perron	Integration order	1%CV	5%CV	10%CV
Residual	-3.39	-3.582	I(0)	2.654	-1.95	-1.602

 Table 6. Granger causality wald tests.

Equation	Excluded	chi2	df	Prob > chi2
Ly	fd	5.5428	1	0.019
Ly	All	5.5428	1	0.019
fd	Ly	1.0707	1	0.301
fd	All	1.0707	1	0.301

academic Journals

Vol. 9(18), pp. 641-660, 28 September, 2015 DOI: 10.5897/AJBM2015.7917 Article Number: C5D59BF55254 ISSN 1993-8233 Copyright © 2015 Author(s) retain the copyright of this article http://www.academicjournals.org/AJBM

African Journal of Business Management

Full Length Research Paper

Dynamics of biotechnology entrepreneurship in South Africa and Brazil

Manessah Alagbaoso*, Teresa Carmichael and Kerrin Myres

Wits Business School, University of Witwatersrand, Johannesburg, South Africa.

Received 3 September 2014; Accepted 22 December, 2014

Biotechnology entrepreneurship is a relatively new and distinct field of entrepreneurship. Most current empirical research is conducted in the developed economies and cannot be directly extrapolated to the developing economies of South Africa and Brazil. This research used a qualitative multiple case studies, in the idiographic philosophical tradition, in two developing economies; South Africa and Brazil. The data collection process included in-depth interviews, documents review and observations, which improved the quality of the research through data triangulation. Ten themes were identified, which formed the basis for developing the proposed theoretical framework. In addition, seven factors that influence the process of biotechnology entrepreneurship in South Africa and Brazil were identified as regulation; funding; infrastructure; skills; entrepreneurial and commercialisation capabilities; market for biotechnology products; and social development. Biotechnology entrepreneurship in Brazil predominantly uses the "system approach" while in South Africa the "individual approach" predominates. The process of biotechnology entrepreneurship in South Africa be "individual approach" predominates the differences in the environmental factors that influence biotechnology entrepreneurship, and management strategies, in these two economies.

Key words: Biotechnology entrepreneurship, entrepreneurship, biotechnology, management strategies, triple helix, university, industry, government, South Africa, Brazil, qualitative analysis.

INTRODUCTION

The introduction of the first biotechnology drug, recombinant insulin, in 1982 marked the turning point in the commercial viability of biotechnology innovation, and its potential to address some of the major global problems of healthcare, food security, energy sufficiency, renewable resources and environmental sustainability (Ahn and Meeks, 2007; Ahn et al., 2010a; Ahn et al., 2010b; Battelle/Biotechnology Industry Organisation, 2010; Ahn and York, 2011; Ahn et al., 2012; Dunham

2012). Within the context of entrepreneurship, biotechnology entrepreneurship is relatively new (Meyers, 2012), research-driven and requires the collaboration of human talent, capital and institutions to achieve economic and social development, job creation, poverty alleviation, skills development and technology transfer. These benefits have captured the interest of the developed and developing economies in programmes and activities aimed at promoting biotechnology entrepreneurship,

*Corresponding author. E-mail: Manessah.Alagbaoso@standardbank.co.za, manessah@telkomsa.net. (+27) 72 922 0206.

Author(s) agree that this article remain permanently open access under the terms of the <u>Creative Commons</u> <u>Attribution License 4.0 International License</u> in order to capitalise on what has been termed the "biocentury" (Battelle/Biotechnology Industry Organisation, 2010).

This study explored the dynamics of biotechnology entrepreneurship in the developing economies of South Africa and Brazil with the aim of understanding the practical realities of biotechnology entrepreneurship in these economies. Although, this research was based on the organizing framework of the individual-opportunity nexus of entrepreneurship (Shane, 2003), the results the individual-opportunity show that nexus of entrepreneurship does not entirely hold for biotechnology entrepreneurship in South Africa and Brazil. Instead, there is a nexus of research and development; and a government-incentivised environment that is conducive for biotechnology entrepreneurship. The main empirical and theoretical contributions of this research include conducting this research in South Africa and Brazil and the development of a proposed theoretical framework of biotechnology entrepreneurship, respectively.

Methodologically, the use of a qualitative multiple case study approach constitutes a methodological contribution. Current studies in biotechnology entrepreneurship make use of a nomothetic philosophical approach and employs quantitative methods within one developing economy context. The policy implications of these dynamics in South Africa and Brazil; as well as implications for the other stakeholders in the biotechnology industry are articulated as being linked to the control of the factors that influence biotechnology entrepreneurship by the various stakeholders. Hence, the implications for government are predominantly linked to regulation and infrastructure; and the implications for the other stakeholders are predominantly linked to funding and skills.

Assumption underpinning the research

There is one assumption that underpins the proposed research. In spite of the difficulties of measuring entrepreneurship in a cross-national context (Carree and Thurik, 1998), it is assumed that the developing economies of South Africa and Brazil are good candidates for the study of biotechnology entrepreneurship in developing economies. There are other countries that could easily be considered for inclusion into this research, such as Malaysia and Indonesia. However, within the context of biotechnology entrepreneurship there are clear similarities that make Brazil and South Africa good candidates for this research. Both countries are classified as efficiency-driven economies in the GEM study (Bosma and Levie, 2010; Kelley et al., 2012) and are seen as ideal emerging market partners for biotechnology alliances with developed economies (Veilleux et al., 2010). In addition, they have fairly developed economy, institutions, educational government institutions. regulatory environment and markets according to the

standard of developing countries (Veilleux et al., 2010). On the basis of patent publication, both have fairly good representation of patent publications according to the standard of developing economies (Bound, 2008; United States Patent and Trademark Office, 2009) and are often included in the Organisation for Cooperation and Development (OECD) analysis and reports on biotechnology (Organisation for Economic Cooperation and Development, 2013d). On the basis of the stage of their biotechnology industry, both have abundant biodiversity and advanced agricultural biotechnology (Cloete et al., 2006; Bound, 2008) and are considered as viable partners for direct foreign investment in biotechnology (Ernst and Young, 2010b). In terms of government policy on biotechnology entrepreneurship, both have high level of government involvement and clearly defined policies on biotechnology development (Department of Science and Technology, 2001; da Silveira and de Carvalho Borges, 2005: Margues and Goncalves Neto, 2007: Bound, 2008; Technology Innovation Agency, 2010; Ernst and Young, 2010b).

LITERATURE REVIEW

The literature on general entrepreneurship recognises the role of the individual in the process of entrepreneurship (Kenney, 1986) in developed and developing economies. The psychological attributes necessary for the individual to function effectively as an entrepreneur include higher levels of cognitive functioning, motivation, leadership qualities, propensity to take risk, action-orientation, selfefficacy, preference for autonomy, self-direction, and differential access to scarce and expensive resources (Shane and Venkataraman, 2000; Eckhardt and Shane, 2003; Shane, 2003). In addition to the psychological attributes, the requisite non-psychological factors such as education and career experience (Barro and Lee, 2000) are necessary for biotechnology entrepreneurship. Studies also recognise the role of the environment or "external forces" (Tushman and Anderson, 1986; Hannan and Freeman, 1987; Acs and Audretsch, 2003) in the process of entrepreneurship in developed and developing economies. In addition, the process of biotechnology entrepreneurship in the developed economies has been seen to be driven primarily through collaboration of stakeholders (Müller et al., 2004), which is similar to having a system of biotechnology entrepreneurship. The literature on the determinants for researchers' choice to engage in commercialisation (Nilsson, Rickne and Bengtsson, 2010) highlights four factors: the perceived role of the university, supportive infrastructure, industrial actor set-up, and networks. The details of these factors include university culture, university infrastructure such as the technology transfer office (TTO), national infrastructure for commercialisation of research, large companies that have receiver capabilities, and

collaboration among key stakeholders. Another component of the literature is the addition of the role of entrepreneurial opportunities (Shane and Venkataraman, 2000). The integration of the individual, environmental and entrepreneurial opportunities (Cunningham and Lischeron, 1991; Shane and Venkataraman, 2000) in exploring the dynamics of biotechnology entrepreneurship has the prospect of providing an in-depth understanding of the key aspects of biotechnology entrepreneurship in South Africa and Brazil.

Biotechnology entrepreneurship

Biotechnology is situated at the boundary of the fields of biology and engineering. It is a combination of science (medical science, biochemistry, molecular biology, cell biology) and industrial production (medical, food, forest industries) (Kivinen and Varelius, 2003), Hence, similar to entrepreneurship, biotechnology has been defined differently by scholars affiliated to either of the underlying fields (Bud, 1991). This contributes to the confusion about what a single unifying definition for the field in the early stages of development should be. However, 21st century biotechnology has been defined as "the use of cellular and biomolecular processes to solve problems or products" useful (Biotechnology make Industrv Organisation, 2008). 20th century biotechnology evolved from an emphasis on population problems and agriculture to a focus on areas such as pharmaceuticals, agricultural chemicals, food production, waste disposal and chemical manufacture (Bud, 1991). In the 21st century, biotechnology has assumed global importance in the areas of healthcare. environmental protection, agriculture. chemistry, and material science (Biotechnology Industry Organisation, 2008), with significant commercial potential (Dibner, 1986; Muller and Fujiwara, 2002; Müller and Herstatt, 2004; Ahn and Meeks, 2007; Ahn et al., 2010a; Ahn et al., 2012).

Biotechnology entrepreneurship in developed economies

In developed economies, the United States (US) leads the chart in biotechnology. The US has more companies, employs more people, invests more in research and development, and earns more than all of Europe combined (Kettler and Casper, 2001). This may be as a result of the early-mover advantage, which the US gained by embracing biotechnology and actively encouraging the development of the industry decades before Europe took the same route. In 2010, the bioscience industry was estimated to have directly created 1.6 million jobs in the US and to be indirectly responsible for about 3.4 million (Battelle/Biotechnology jobs in total Industry Organisation, 2012), with a net income of \$3.7 billion for

publicly traded biotech companies in 2009 (Ernst and Young, 2010b). Owing to globalisation and through international collaboration, there is extensive interaction between the American and European biotechnology industry through the big multinational biotechnology corporations (Dibner, 1986) and research scientist networks. At a governmental level, many European countries, and an Asian country such as Japan, implement similar policy initiatives to America in order to fast-track their biotechnology industries. The existence of similar policy initiatives to those of America has been confirmed in the biotechnology industries of Finland (Kivinen and Varelius, 2003); the United Kingdom, Germany and Japan (Dibner, 1986; Müller, 2002; Muller and Fujiwara, 2002; Müller et al., 2004); and Sweden (Nilsson, 2001). The combination of American and European biotechnology industries, under the banner of developed economies, constitutes a near total domination of the entirety of the global biotechnology industry, with the developing economies having very little or no current impact. It is difficult to make a direct comparison between the biotechnology industry in the developed and developing economies, except in cases such as the biofuel industry in Brazil, owing to the general lack of empirical research and data, and the undeveloped nature of the industry in most developing economies. However, the GEM report provides a basis of comparison for general entrepreneurial activities across these two types of economies (Herrington et al., 2008; Bosma and Levie, 2010), which is assumed to provide a similar comparative basis for biotechnology entrepreneurship.

The GEM report has categorised the national conditions of the developed economies under "innovationdriven" as shown in Table 1. For developed economies, the basic requirements (institutions, infrastructure, macroeconomic stability, health and primary education) and efficiency enhancers (higher education and training, goods market efficiency, labour market efficiency, financial market sophistication, technological readiness and market size) are in place and are maintained. Nevertheless, the key focus is on the entrepreneurial conditions (Bosma and Levie, 2010) such as entrepreneurial finance, government policies, government entrepreneurship programmes, entrepreneurship education, R&D transfer, commercial and legal infrastructure for entrepreneurship, internal market openness, physical infrastructure for entrepreneurship, and cultural and social norms. The prevalence of entrepreneurial activities in the US attests to the availability of basic requirements and efficiency enhancers. About 4 to 6% of America's working population take action to start a new business annually, and about 40% experience bouts of selfemployment in their lifetime (Acs and Audretsch, 2003:5). The result of this is that an estimated 761,000 new corporations, the number of which increases to 4.5 million with the inclusion of all forms of business, are started per year in America as at 1998 (Ács and Audretsch, 2003:28).

Table 1. Importance of different types of national conditions for economic development (Bosma and Levie, 2010).

	Basic requirements	Efficiency enhancers	Entrepreneurial conditions
Factor-driven economies	Key focus	Develop	Start enabling
Efficiency-driven economies	Maintain	Key focus	Develop
Innovation-driven economies	Maintain	Maintain	Key focus

Table 2. Differences in national conditions for developed and emerging economies (Adapted from (Phan et al., 2008).

Measure	Developed economies	Emerging economies
Competitiveness	High competitive intensity	Vagaries of policy making and not much competition
Socio-economic linkages	These issues are usually hidden or taken for granted in extant literature	Link between economic development, social welfare and entrepreneurial action
Inputs	Availability of critical mass of inputs required to ignite entrepreneurial action	Lacks the critical mass of inputs (capital, human talent, technology) required to ignite entrepreneurial action
Nature of entrepreneurial action	May be accidental and spontaneous as the necessary conditions already exist	Non-accidental and purposefully orchestrated by government, providing resource endowments, institutions and markets. Government provide both macro- and micro-economic factors aimed at providing incentives for entrepreneurial action

The trend for the rest of the developed economies is expected to be similar to America, although not to the same magnitude. Some of the differences linked to the national conditions for the developed and developing economies (Phan, Venkataraman and Velamuri, 2008) are shown in Table 2. Biotechnology entrepreneurship activities are currently entrenched in the developed economies, and are continually being improved as biotechnology is expected to be the economic growth engine of the 21st century (Battelle/Biotechnology Industry Organisation, 2010). As might be expected, most of the published literature on biotechnology entrepreneurship is based in the developed economies, where industry statistics have been tracked for decades. The empirical studies on biotechnology entrepreneurship for the developing economies are still at the exploratory stage and the industry has not developed to a stage where statistics are readily available and tracked.

Biotechnology entrepreneurship in emerging economies

In developing economies, biotechnology entrepreneurship holds the tantalising prospect of significantly contributing to food security, improved agricultural output, sustainable environmental development practices, improved healthcare, job creation, poverty alleviation and economic development (Clarke, 2002). The achievement of all, or any, of these benefits depends on the national conditions that exist in the developing economies. The "bigger" economies of the developing world, such as BRIC (Brazil, Russia, India and China), South Africa and possibly Nigeria (Department of Science and Technology, 2001), may be in a better position to exploit the benefits of biotechnology in a globalised world, whether through technology transfer, innovative development of the industry, development of particular niches within the biotechnology industry or a combination of these and other options. These bigger economies represent a vital link between the developing and the developed economies as the basic requirements already exist in these economies (Fontes 2001). The GEM report (Kelley, Singer and Herrington, 2012) classifies South Africa and Brazil as efficiency-driven and as such the key focus is on the efficiency enhancers such as higher education and training, goods market efficiency, labour market efficiency, financial market sophistication, technological readiness, market size. While, these efficiency enhancers are fundamental to the development of entrepreneurial culture in general, the developing economies still need to develop the entrepreneurial conditions necessary for an innovation-driven industry such as biotechnology (Herrington et al., 2008; Bosma and Levie, 2010).

The global issues of human health, food security, renewable resources and environmental sustainability (Battelle/Biotechnology Industry Organisation, 2010) that are addressed by biotechnological solutions are more prevalent in developing economies as are the issues of economic and social development, unemployment and



Figure 1. The individual-opportunity nexus framework (Shane, 2003:11).

global competitiveness. These issues highlight the importance of understanding and developing entrepreneurship in general, and specifically biotechnology entrepreneurship, in the emerging economies.

Research question

The research question explored in this study is:

Question 1

How do bioscientists carry out biotechnology entrepreneurship in the developing economies of South Africa and Brazil?

Organising framework

The individual-opportunity nexus framework of entrepreneurship (Shane, 2003:11) (Figure 1) was used as the organising framework for this research, which guided the exploration of the dynamics of biotechnology entrepreneurship in the developing economies of South Africa and Brazil. Shane's (2003) framework considers the existence of entrepreneurial opportunities as independent of the actors and as needing to be discovered by enterprising individuals. The individual attributes needed to exploit these opportunities effectively include psychological factors, such as cognition and motivation and non-psychological factors such as education and career experience. In the environment of entrepreneurship, the three categories of factors believed to influence productive entrepreneurial activity are the economic, political and cultural environments (Shane, 2003). There are marked differences between the developed and developing countries in all three categories of environmental factors. While, the four aspects of the economic environment: wealth, economic stability, capital availability and taxation, are all at levels and favourable advanced for productive entrepreneurship in the developed economies, the developing economies face issues of poverty, economic instability, lack of capital and restrictive tax laws (Herrington et al., 2008; Bosma and Levie, 2010). Similarly, political instability in developing economies and low levels of a national culture of innovation and entrepreneurship hamper productive entrepreneurial activity (Herrington et al., 2010).

Shane (2003) individual-opportunity nexus framework for entrepreneurship is able to explain the process of biotechnology entrepreneurship as the literature on biotechnology entrepreneurship aligns to the components of the individual-opportunity nexus framework (Pisano 1990; 1991). This makes this framework useful for understanding the process of biotechnology entrepreneurship in South Africa and Brazil. The process of biotechnology entrepreneurship in South Africa and Brazil is then compared to the literature on the process of biotechnology entrepreneurship in developed economies, in order to identify similarities and differences.

RESEARCH METHODOLOGY

This research was conducted in the idiographic philosophical tradition. The rationale for using this is the complex nature of the social interactions involved at the individual, organisational, institutional and national levels which underlie the dynamics of biotechnology entrepreneurship. These interactions are non-linear and have multiple overlapping meanings, and are suitably studied through the case study method. Primary data from a case study design was used for this study. Multiple cases, comprising South Africa and Brazil, were investigated in order to achieve literal and theoretical replication (Eisenhardt, 1989, 1991; Yin, 2009). The individual cases were holistic within the multiple-case study. The case study was carried out at country level of analysis with interviews conducted with individuals associated with the biotechnology industry in various capacities. Environmental and institutional factors have key influences on the dynamics of biotechnology entrepreneurship at a country level. The use of a country level of analysis was expected to embody these environmental factors and be broad enough to provide a true understanding of the dynamics of biotechnology entrepreneurship in the developing economy context. In addition, conducting interviews with individuals within the biotechnology industry ensured that the lived dynamics of biotechnology entrepreneurship within its original context were adequately captured (Creswell, 2009; Yin, 2009).

Data analysis

The qualitative data analysis for this research was conducted in the tradition of building theory from case studies in order to understand the "how" and "why" of contemporary events (Eisenhardt and Graebner, 2007; Yin, 2009:8; Klonoski, 2013). The dynamics of biotechnology entrepreneurship in South Africa and Brazil were examined in detail to generate theoretical insights (Welch, Piekkari, Plakoyiannaki and Paavilainen-mäntymäki, 2011), understand their peculiarities (Stake, 2006), and the structures and contexts (Klonoski, 2013) in which observed behaviours were analysed. A Computer Aided Qualitative Data Analysis Software (CAQDAS) package called Atlas.ti version 7.1.6 was used for the data analysis. The use of Atlas.ti was meant to aid rapid, consistent and rigorous qualitative data analysis (Weitzman, 1999; Rambaree, 2007; Hwang, 2008), and extended the researcher's ability to organise, remember and be systematic (Zdenek, 2008).

FINDINGS AND DISCUSSION

The cross-case analysis is an aggregate view (Stake, 2006) of the research on the dynamics of biotechnology entrepreneurship in South Africa and Brazil. The patterns that emerged from the peculiarities of the individual case studies are aggregated to themes and contrasted with the literature on biotechnology entrepreneurship to enable an accurate interpretation towards developing theory. The patterns identified from the within-case analyses of South Africa and Brazil are presented in Table 3 below. The patterns that emerged from within the case analyses of South Africa and Brazil (Table 3) resulted in 14 areas of alignment between South Africa and Brazil, from which the emerging themes were derived. The patterns referred to the same concept but not necessarily with the same experience or outcome. Hence, there were instances of positive versus negative experiences across the patterns.

The patterns which were a function of the other patterns; and the patterns that appeared in only one of the cases were not considered for inclusion as themes.

Theme 1

"There is a dual approach to the practice of biotechnology entrepreneurship, which can be seen as a 'system' or as an 'individual"

The manifestation of theme 1 resulted in dual approaches to the process of biotechnology entrepreneurship in South Africa and Brazil. With the availability of infrastructure and support structures, as experienced by the respondents in Brazil, the "system" approach to biotechnology entrepreneurship was adopted. On the other hand, the lack of infrastructure and support structures, as experienced by the respondents in South Africa, manifested in the adoption of the "individual" approach to biotechnology entrepreneurship. In the "system" approach to biotechnology entrepreneurship, a bioscientist may not necessarily need to acquire the entrepreneurial and business management skills necessary to manage the commercialisation of research successfully. The "system" delivers the entrepreneurial and commercialisation resources required for effectively carrying out biotechnology entrepreneurship, through the government, venture capitalists, large biotechnology companies, or a combination of these. This form of biotechnology entrepreneurship is manifested in Brazil through the commercialisation of research that involved the provision of the entrepreneurial and commercialisation resources by large biotechnology companies and government. This is further manifested in the availability of incubators and technology parks and the effective collaboration that provides the other support structures needed for the effective commercialisation of research.

The bioscientists who participated in this research in Brazil did not need to go through formal training to acquire the entrepreneurial and commercialisation skills necessary for the commercialisation of their research. The system delivered this capability and they were upskilled in commercialisation through practical experience. The "individual" approach entails bioscientists acquiring the entrepreneurial and commercialisation skills that enable them to manage the commercialisation of research successfully. This form of biotechnology entrepreneurship develops as a result of the lack of the support structures and infrastructure to provide the entrepreneurial and commercialising resources. This form of biotechnology entrepreneurship is manifested in South Africa due to the lack of a developed venture capital industry, the large established biotechnology companies and other critical infrastructure and support structures. The bioscientists acquire the necessary entrepreneurial and commercialisation skills through formal training and

 Table 3. Patterns from within-case analyses of South Africa and Brazil.

Emerging themes	Patterns in South Africa	Patterns in Brazil
Individual versus system approach	Owing to lack of adequate support system to provide the entrepreneurial and commercialisation skills needed to commercialise their research, these bioscientists are often forced to acquire entrepreneurial and commercialisation skills to commercialise their research effectively. Consequently, their practice of biotechnology entrepreneurship is as individuals	The availability of infrastructure and support structures, which provides the entrepreneurial and commercialisation skills, enables the bioscientists to employ the system approach to biotechnology entrepreneurship
Collaboration	Most of the respondents believe that there is a lack of collaboration among key stakeholders in biotechnology entrepreneurship in South Africa. Although most of the respondents experience evidence of collaboration with key stakeholders, they agree that a need exists for effective collaboration among key stakeholders	There is collaboration among the key stakeholders and this collaboration is acknowledged to be important to the effective development of biotechnology entrepreneurship in Brazil
The role of government	The lack of direction from the government, through its implementation agencies, is consistently deemed to be a constraint by most of the respondents and their activities deemed to be hindering the development of the biotechnology industry	The government plays an important role in incentivising bioentrepreneurial activities in Brazil
Regulation	The regulatory environment is consistently deemed to be a constraint by most of the respondents	The regulatory environment for biotechnology entrepreneurship in Brazil is deemed to be unfavourable
Funding	The lack of appropriate funding is consistently deemed to be a constraint by most of the respondents	There is availability of appropriate funding for the development of biotechnology entrepreneurship in Brazil
Markets for biotechnology products	The lack of developed markets for biotechnology products is consistently deemed to be a constraint by most of the respondents	The local market for biotechnology products is considered developed
The skills required for biotechnology entrepreneurship	Although most of the respondents believe that the scientific and research skills needed for biotechnology entrepreneurship are available in South Africa, they also deem the entrepreneurial and commercialisation skills to be lacking	There is an availability of the research and scientific skills necessary for the development of biotechnology entrepreneurship. The entrepreneurial and commercialisation skills are available through the system of biotechnology entrepreneurship in Brazil
Nature of bioentrepreneurial opportunities	The bioentrepreneurial opportunities were identified predominantly in the area of problem opportunities, followed by efficiency opportunities and a small instance of innovation opportunities	The bioentrepreneurial opportunities were identified predominantly in the area of efficiency opportunities, followed by innovation opportunities and a small instance of problem opportunities
Biodiversity	The abundance of biodiversity is deemed to be a positive factor by most of the respondents in the development of biotechnology entrepreneurship in South Africa	The abundant biodiversity constitutes a source of competitive advantage for the development of the biotechnology industry in Brazil
R and D is the defining step in biotechnology entrepreneurship in South Africa and Brazil	The discovery of these bioentrepreneurial opportunities requires R&D. The intensity and the R&D spend in South Africa are deemed to be below the levels of the developed economies	The intensity and the R&D spend in Brazil are deemed to be adequate and are being improved by the government and the private sector. However, they are below the levels of the developed economies
Exploitation of bioentrepreneurial opportunities effected through collaboration	The exploitation of these bioentrepreneurial opportunities requires a strategic alliance of research institutions, venture capitalists, large established biotechnology companies, and government. This process in South Africa lacks the key stakeholders: venture capitalists and large established biotechnology companies	This process in Brazil has all the key stakeholders: venture capitalists, research institutions, government and large established biotechnology companies

Table 3. Contd.

Commercialisation of research is determined by the dynamics of the earlier steps in the process of biotechnology entrepreneurship	The commercialisation of research in South Africa utilises both firm formation and licensing and is deemed to be inefficient due to a lack of entrepreneurial and commercialisation skills	The commercialisation of research in Brazil utilises both firm formation and licensing and is deemed to be efficient due to the efficiencies in the earlier parts of the biotechnology value chain, and the availability of the system to enable the commercialisation of research
Country competitiveness	The emphasis on country competitiveness is consistently deemed by most of the respondents to be an important consideration for biotechnology entrepreneurship in South Africa	Global competitiveness is a key factor in the approach to biotechnology entrepreneurship in Brazil
Environment of biotechnology entrepreneurship	The general environment for the development of biotechnology entrepreneurship in South Africa is deemed by most of the respondents to be unfavourable	There is an overall conducive environment for the development of biotechnology entrepreneurship in Brazil
	Most of the respondents agree that there is a misalignment between the requirements for being an academic and being an entrepreneur. In addition, the university culture is deemed to mostly prioritise publication over commercialisation	
	The venture capital industry is consistently deemed to be underdeveloped in South Africa	
		The high cost of funding in Brazil constrains the development of the biotechnology industry
		The universities in Brazil are good at conducting the necessary research required for the development of biotechnology entrepreneurship in Brazil
		The politicians do not have a full understanding of the requirements of the biotechnology industry
		The aggregate capacity for innovation needs to be improved
		The inefficiencies of the government and its agencies affect the effectiveness of the development of biotechnology entrepreneurship in Brazil

approach the process in their individual capacity as is often the case in general entrepreneurship. The bioscientists who participated in this research in South Africa needed formal training in business management to acquire the entrepreneurial and commercialisation skills needed for effective commercialisation of their research. This training took the form of a Master of Business Administration (MBA), Management Advancement Programme (MAP), and entrepreneurial skills development programme run by some of the stakeholders and sponsored by the government.

Theme 2

"Biotechnology opportunities occur in the form of problem, efficiency, and innovation opportunities"

Most of the respondents in South Africa see bioentrepre-

neurial opportunities in the areas of biodiversity and problems related to diseases and food security. These types of opportunities, linked to problems of diseases, food security, the environment and energy are designated as "problem opportunities" (researcher's The biodiversity highlighted by the synthesis). respondents falls within the second category of opportunities. The second form of bioentrepreneurial opportunities identified by the respondents is designated as "efficiency opportunities", which mostly occur in the areas of bioprocessing and biomanufacturing. While, the efficiency opportunities may not be as ubiquitous as problem opportunities in South Africa due to the low level of industrialisation, the realised efficiencies do provide the opportunity to solve some of the problems of food security and environmental sustainability, and create populations energy sufficiencv healthier and (Biotechnology Industry Organisation, 2008).

In Brazil, the bioentrepreneurial opportunities are

mostly aligned to efficiency and innovation opportunities and a component of problem opportunities. The efficiency opportunities are manifested mostly in the areas of biodiversity; bioenergy; genomics; pharmaceuticals; bioprocessing and biomanufacturing. The realised efficiencies do provide the opportunity to solve some of the problems of food security and environmental sustainability, and achieve healthier populations and energy conservation. The innovation opportunities are manifested in the areas of bioenergy; genomics; vaccines; antibiotics; and the use of indigenous knowledge and biodiversity. The nature of the bioentrepreneurial opportunities in Brazil means that it straddles the opportunity continuum between the developing and developed economies, with problem and efficiency opportunities aligning it to the characteristics of developing economies; and innovation opportunities aligning it to the characteristics of developed economies. Hence, the different types of bioentrepreneurial opportunities are designated as problem, efficiency, and innovation opportunities (researcher's synthesis).

The entrepreneurial opportunities in biotechnology seem be different from the entrepreneurial to opportunities in general entrepreneurship in some key aspects. While, general entrepreneurial opportunities are not known in advance and require enterprising individuals with special psychological attributes (McClelland, 1961; Kihlstrom and Laffont, 1979; Schere, 1982; Gartner, 1990) to uncover them, bioentrepreneurial opportunities are mostly known in advance and require bioscientific skills to obtain a solution (Müller et al., 2004). The exploitation of general entrepreneurial opportunities is also mostly dependent on the individual entrepreneurs and their organising abilities (Shane, 2003). However, the exploitation of bioentrepreneurial opportunities mostly requires a strategic alliance involving government, large established companies, venture capitalists, and research institutions (Liebeskind, Oliver, Zucker and Brewer, 1996; Audretsch and Stephan, 1998; Agrawal, 2001; Shane, 2003; Müller et al., 2004; Rothaermel and Deeds, 2004; Powell, White, Koput and Owen-Smith, 2005; Ahn and Meeks, 2007; Sytch and Bubenzer, 2008).

Theme 3

"Regulation is a critical factor in biotechnology entrepreneurship in South Africa and Brazil"

The regulatory environment of biotechnology entrepreneurship in South Africa was deemed to be nonconducive for biotechnology entrepreneurship by the respondents. This regulatory challenge was also recognised in the new bio-economy strategy to be implemented by the government of South Africa.

Similarly, the experience of challenges with the regulatory environment for biotechnology entrepreneurship in Brazil was unanimous across all the transcripts. The specific areas that were highlighted as challenges are policies related to intellectual property ownership; policies related to technology transfer from the university to the industry; taxation laws; and labour laws. The regulatory environment of biotechnology entrepreneurship in South Africa and Brazil defines the existence of the industry, the rules of engagement among the stakeholders and the general environment in which the industry operates. Hence, the regulatory environment is important for general entrepreneurship (Shane and Venkataraman, 2000; Eckhardt and Shane, 2003; Shane, 2003; Lingelbach, De La Vina and Asel, 2005; Phan, Venkataraman and Velamuri, 2008) and determines the effectiveness of the process of entrepreneurship to some extent. The regulatory environment is also considered to be one of the primary requirements for effective R and D, in addition to research universities, other research institutions, a developed scientific educational curriculum, a national culture that supports scientific endeavour, and talented individuals (Organisation for Economic Cooperation and Development, 2009). The GEM also highlights the regulatory environment as one of the differentiators of the efficiency-driven and innovationdriven countries (Kelley et al., 2012) in relation to the effectiveness of the process of entrepreneurship. In comparison to general entrepreneurship, the policy and regulatory environment plays a bigger role in biotechnology entrepreneurship in South Africa and Brazil due to the multifaceted nature of biotechnology (Dibner, 1986; Muller and Fujiwara, 2002; Müller et al., 2004; Ahn and Meeks, 2007; Biotechnology Industry Organisation, 2008; Battelle/Biotechnology Industry Organisation, 2012). Biotechnology impacts on critical areas of the societv such as food security, human health, environmental sustainability and energy sufficiency (Dibner, 1986; Muller and Fujiwara, 2002; Müller et al., 2004; Ahn and Meeks, 2007; Biotechnology Industry Organisation, 2008; Battelle/Biotechnology Industry Organisation, 2012).

Theme 4

"An overall conducive environment is necessary for the effective development of biotechnology entrepreneurship in South Africa and Brazil"

overall environment of biotechnology The entrepreneurship in South Africa presents many challenges according to the respondents. The challenges highlighted by most of the respondents include a policy and regulatory environment that is not conducive to entrepreneurship; lack of appropriate funding; an inclination by the universities to prioritise publication over commercialisation of research; lack of government leadership and direction; lack of aggregate skills, as well as entrepreneurial and commercialisation skills; and lack of a developed market for biotechnology products. These

challenges were corroborated by the document analysis, which highlighted areas of challenge such as human capital, knowledge exploitation, market development and governance. Of the 19 areas highlighted only one was mostly seen as not a challenge while the remaining 18 areas were seen as challenges by most of the respondents.

In relation to Brazil, the environment of biotechnology entrepreneurship, as experienced by the respondents, is mixed with an equal occurrence of key challenges and areas considered not to represent gaps in the development of biotechnology entrepreneurship in Brazil. There is an equal split between the factors designated as favourable and those seen as challenges. Of the 14 areas highlighted seven were highlighted as gaps, and seven, not as gaps. The challenges include the regulatory environment, especially related to taxation; shortage of aggregate skills; national culture of seeing entrepreneurship as impure; high interest rate, which results in high cost of funds; bureaucracy; corruption and politics.

On the other hand, the areas considered not to represent challenges are the provision of direction and leadership by the government; availability of appropriate funding; the size of the market for biotechnology products in Brazil; availability of relevant infrastructure; a university culture that is conducive to the development of biotechnology entrepreneurship; availability of adequate capacity for biotechnology entrepreneurship; and good scope for international collaboration. The institutional environment needed for supporting the development of entrepreneurship in general (Kelley et al., 2012; Urban, 2013) is also applicable to biotechnology entrepreneurship. The three categories of factors believed to influence productive entrepreneurial activity are the economic, political and cultural environments (Shane, 2003).

However, the overall environment for the development of biotechnology entrepreneurship in South Africa and Brazil includes regulatory and social environments. The regulatory environment is particularly important in the environment of biotechnology entrepreneurship (Müller et al., 2004; Ahn and Meeks, 2007). Also, the innovation and entrepreneurial environment proposed by Herrington et al. (2012) contains key elements that are important to the development of biotechnology entrepreneurship, such as entrepreneurial finance, government policies, government entrepreneurship programmes, entrepreneurship education, R and D transfer, commercial and legal infrastructure for entrepreneurship, internal market openness, physical infrastructure for entrepreneurship, and cultural and social norms.

Theme 5

"Research and development is the defining step in bio-

technology entrepreneurship in South Africa and Brazil"

According to the OECD (2009), the primary requirements for R and D to be effective include research universities, other research institutions, a developed scientific educational curriculum, a national culture that supports scientific endeavour, a favourable regulatory environment and talented individuals. Most of the respondents in South Africa deem the funding for R and D to be inadequate and see it as a gap in the effort to develop the biotechnology industry in South Africa.

By contrast, R and D in Brazil is well supported and funded through the national and state government agencies; private sector initiatives; international collaboration; and large biotechnological companies. In spite of this, the R and D spend still falls below the OECD average and the respondents believed that more needed to be done to sustain the level of research intensity in the industry.

Opportunity discovery in general entrepreneurship is different from opportunity discovery in biotechnology entrepreneurship. Opportunity discovery in biotechnology entrepreneurship is more of a "creation" (Audretsch et al., 2008) than a "discovery" (Alvarez and Barney, 2007) as the solution to exploit the opportunities is created or discovered through R and D.

R and D is a key step that differentiates biotechnology entrepreneurship from general entrepreneurship. This is also the key step that contributes to the predominance of non-psychological factors such as education and career experience (Barro and Lee, 2000) in biotechnology entrepreneurship. Given that the discovery of biotechnological opportunities requires R and D, it simply follows that the direct consequence of low R and D spend and activities in developing economies is low levels of biotechnological activities.

Theme 6

"There are four types of skills required in biotechnology entrepreneurship in South Africa and Brazil, namely: research, scientific, entrepreneurial, and commercialisation skills"

The importance of skills to the development of biotechnology entrepreneurship in South Africa was highlighted by most of the respondents. In analysing the experience of the respondents regarding the skills landscape, the four broad categories of skills highlighted by the respondents were scientific, research. entrepreneurial, and commercialisation skills. While most of the respondents believed that scientific and research skills are available in South Africa, the entrepreneurial and commercialisation skills were deemed to be lacking. The respondents in Brazil believed that the research and scientific skills are available at individual levels for the

development of biotechnology entrepreneurship in the country. However, the aggregate skills were deemed to be inadequate and as negatively impacting on the speed and scale of the development of the industry.

The lack of entrepreneurial and commercialisation skills was identified in prior studies on the biotechnology industry in South Africa in the form of a lack of skills (Lingelbach et al., 2005), and low levels of commercialisation of biotechnology products (Cloete et al., 2006). This distinction between the types of skills that are lacking is important given the general discourse on the low level of mathematics and science education in South Africa (Department of Science and Technology, 2001; Department of Science and Technology, 2007), which can easily be wrongly interpreted to mean a lack of and research science skills for biotechnology entrepreneurship in South Africa.

There have been much emphasis on the importance of skills in biotechnology entrepreneurship across industry research; empirical research on developed economies; and empirical research on developing economies (Department of Science and Technology, 2001; Lingelbach et al., 2005; Ahn and Meeks, 2007; Department of Science and Technology, 2007; Phan et al., 2008; Ahn et al., 2010a; Ahn et al., 2010b; Battelle/Biotechnology Industry Organisation, 2010; Ahn and York, 2011; Ahn et al., 2012; Dunham et al., 2012; Kelley et al., 2012).

In contrast to general entrepreneurship, the research and scientific skills are must-haves for participation in biotechnology entrepreneurship, as R and D is used to discover or create opportunities in biotechnology entrepreneurship (Ahn and Meeks, 2007; Phan et al., 2008; Ahn et al., 2010a; Ahn et al., 2010b; Battelle/Biotechnology Industry Organisation, 2010; Ahn and York, 2011; Ahn et al., 2012; Dunham et al., 2012).

Theme 7

"Effective collaboration among key stakeholders is important to the development of biotechnology entrepreneurship in South Africa and Brazil"

The key manifestation of the importance of effective collaboration among the stakeholders in biotechnology entrepreneurship in South Africa and Brazil relates to collaborative projects, which resulted in positive outcomes in Brazil, where the collaboration among the stakeholders was deemed to be effective. In South Africa, where the collaboration among the stakeholders was deemed to be ineffective, few successful collaborative projects were achieved.

There was no successful collaborative research project in South Africa highlighted by the South African respondents; and a few successful collaborative research projects in South Africa were highlighted in the literature on South Africa (Cloete et al., 2006). This may be as a result of the ineffective collaboration among the key stakeholders.

The relevance of the interactions among the university, industry and government, in transforming academic research into societal and economic capital, is demonstrated in the field of biotechnology (Liebeskind et al., 1996; Agrawal, 2001; Müller et al., 2004; Rothaermel and Deeds, 2004; Powell et al., 2005; Ahn and Meeks, 2007; Sytch and Bubenzer, 2008). The collaboration these three stakeholders (Etzkowitz and among Leydesdorff, 1997; Leydesdorff and Etzkowitz, 1998, 2001) has recently been considered to involve a fourth stakeholder (Afonso, Monteiro and Thompson, 2010; Marcovich and Shinn, 2011; Afonso, Monteiro and Thompson, 2012; Leydesdorff, 2012) leading to a quadruple helix. The quadruple helix is context-specific and in the case of the developing economies the socioeconomic linkages may point to the possibility of the fourth stakeholder being the society.

The triple helix relations in South Africa and Brazil, based on the experience of the respondents in this research, do not follow a sequence of triple helix I, II and III as postulated by Etzkowitz and Leydesdorff (2000). In the model, in the progression from triple helix I to III the influence and control of the government is diminished. However, it is envisaged that the influence and control of the government will not diminish in the biotechnology industry in South Africa and Brazil. This leads to a scenario where there is a possibility of a hybrid model of Triple Helix I, which represents a configuration in which the government encompasses both industry and university and directs the interaction and relations between them; and Triple Helix III, in which overlapping institutional spheres generate a knowledge infrastructure, with overlapping roles and hybrid organisations emerging at the interfaces (Etzkowitz and Leydesdorff, 2000) without direct control by any of the institutions.

The uniqueness of this hybrid model is that the gains of Triple Helix III will be appropriated in an environment where the government maintains the influence and control over the interactions of the stakeholders. It is also noteworthy that this bypasses the Triple Helix II, in which the three institutional spheres are separate with strong borders dividing them and restricted relations (Etzkowitz and Leydesdorff, 2000). In support of this assertion is that there is little chance of separation from the government, given its importance and the needs of the industry.

Theme 8

"The government plays an important role in biotechnology entrepreneurship in South Africa and Brazil"

In Brazil, where government leadership and direction were deemed to be effective there have been notable

successes in biotechnology entrepreneurship. The challenges to biotechnology entrepreneurship in Brazil identified by most of the respondents were also mostly in areas of government responsibility such as the policy and regulatory environment; bureaucracy and inefficiencies of the government agencies; high cost of funding; lack of aggregate skills; politics; and corruption.

By contrast, the respondents in South Africa highlighted the lack of government leadership and direction as one of the challenges in the environment of biotechnology entrepreneurship. The government agencies tasked with the implementation of the policies and strategies for biotechnology entrepreneurship were deemed to be ineffective by most of the respondents. Hence, all but one of the 19 areas highlighted by the respondents as impacting on the environment of biotechnology entrepreneurship in South Africa were designated as challenges.

The role of the government in biotechnology entrepreneurship is to provide an enabling regulatory environment and act as a facilitator (Fontes, 2001; Müller et al., 2004; Ahn and Meeks, 2007; Nilsson et al., 2010). The regulatory environment includes the policies and laws that impact on biotechnology entrepreneurship in South Africa and Brazil, such as the national bioeconomy strategy of South Africa; the national biotechnology development policy of Brazil; the intellectual property policy; the technology transfer policy; the taxation laws; and labour laws. The role of the government as a facilitator includes creating a favourable environment for biotechnology entrepreneurship and providing grants and incentives (Müller et al., 2004).

These roles as articulated in the literature relate to the developed economies, and specifically to Germany and Japan (Müller et al., 2004). This is corroborated by the role of the government in the biotechnology industry of the US (Ahn and Meeks, 2007) and the rest of the developed economies. Few studies on the role of the government in biotechnology entrepreneurship are specific to the developing economies. However, from a general entrepreneurship point of view, the nature of entrepreneurial action in the developing economies is purposefully orchestrated by non-accidental and government, providing resource endowments, institutions and markets. Government provides both macro- and microeconomic factors aimed at providing incentives for entrepreneurial action (Phan et al., 2008).

Theme 9

"Funding is a critical factor in biotechnology entrepreneurship in South Africa and Brazil"

The funding environment for biotechnology entrepreneurship in South Africa was deemed unfavourable by the respondents. The lack of a developed venture capital industry and large biotechnology companies meant that the government was deemed to be the only source of funding for biotechnology entrepreneurship in South Africa. This presented another challenge in that the funding from the government is often not appropriate for the nature of the biotechnology industry in terms of risk profile and timeframe.

By contrast, the respondents in Brazil were of the opinion that appropriate funding was available in the biotechnology industry in Brazil. The sources of this funding are primarily the government through its national and state funding agencies; the private sector; and venture capitalists.

The availability of capital (Shane, 2003) is an important factor in any entrepreneurial activity and biotechnology entrepreneurship is often associated with both government and venture capital funding sources (Audretsch, Taylor Aldridge and Perry, 2008).

The experience of the bioentrepreneurs in South Africa was mostly restricted to the government source of funding, owing to the lack of a developed venture capital industry for biotechnology (Audretsch et al., 2008). This situation may be unique to the South African biotechnology industry, as venture capital funding is a core component of biotechnology entrepreneurship in developed economies (Ahn and Meeks, 2007; Ahn et al., 2010a).

The availability of appropriate funding in Brazil through these sources is similar to the process of biotechnology entrepreneurship in developed economies (Ahn and Meeks, 2007; Ahn et al., 2010a), where the government plays a big role despite the availability of other sources of funding.

Theme 10

"The local market for biotechnology products is a critical factor in biotechnology entrepreneurship in South Africa and Brazil"

The size of the market for biotechnology products in South Africa and Brazil was seen as a critical factor in biotechnology entrepreneurship. In South Africa, the size of the market was deemed small and underdeveloped by most of the respondents. From a medical diagnostics point of view, there seems to be an adequate market in South Africa. This may be due to the nature of the products and the target market.

The government was seen as the biggest buyer of biotechnology products in South Africa and, hence, influences the size of the market. The ability of the local companies to sell to the government determines their success in the local market. If the local biotechnology companies cannot win the competition in their local market, the odds are heavily against them making inroads in the international market with competitors with a far better operational environment than in South Africa. By contrast, the respondents believe that the Brazilian market for biotechnology products is developed. The emphasis on national priorities and solving the problems of the country means that the areas of focus, such as agriculture, environment, industry and healthcare, are aligned to the needs of the country. Hence, the local demand creates a local market for the biotechnology products.

Despite having a developed market for biotechnology products, the need to compete internationally was expressed by the respondents as one of the factors necessary for the development of the industry in Brazil. The competitive pressure highlighted by the respondents in their local market was from biotechnology companies outside the borders of their local market, in both developing and developed economies. Hence, biotechnology entrepreneurship is global; the choice of market is mostly determined by the type of bioentrepreneurial opportunity that is being exploited by the bioentrepreneur; and a developed local market may be a factor in favourably positioning the bioentrepreneur to compete globally (Zylberberg et al., 2012).

The proposed theoretical framework of biotechnology entrepreneurship in South Africa and Brazil

The organising framework that guided this research framework (Shane. 2003) is а for general entrepreneurship and is not specific to biotechnology entrepreneurship. The existing literature on biotechnology entrepreneurship that has a framework related to the process of biotechnology entrepreneurship is that of Müller et al. (2004), on the study of sources of bioentrepreneurship in Germany and Japan. However, the model by Müller et al. (2004) is restricted to the stakeholders needed for the exploitation of bioentrepreneurship opportunities. Two previous empirical studies on biotechnology in South Africa (Gastrow, 2008) addressed the quantitative profile of biotechnology research and development in South Africa and the state of biotechnology in South Africa, with emphasis on the national biotechnology strategy and its implementation, respectively. Other industry studies on the biotechnology industry in South Africa have concentrated on the key initiatives driven by the government and the performance of the industry (Ernst 2006; Organisation for Economic and Young, Cooperation and Development, 2009; Ernst and Young, 2010a; 2010b; Organisation for Economic Cooperation and Development, 2013b; 2013a). No current empirical or industry study has addressed the process of biotechnology entrepreneurship in South Africa by the bioentrepreneurs and stakeholders engaged in the process. Previous studies on biotechnology entrepreneurship in Brazil have been more extensive in addressing the biotechnology industry holistically (Zylberberg et al., 2012). However, a language limitation prevented the researcher from being able to review studies in Brazil that were published in Portuguese. There is no existing theoretical framework of biotechnology entrepreneurship that enables the study and understanding of the process of biotechnology entrepreneurship in different contexts, whether in the developed or developing economies. The themes identified in the cross-case analysis of South Africa and Brazil has characteristics that differentiate the process of biotechnology entrepreneurship from the process articulated in the organising framework of the individualopportunity nexus of entrepreneurship.

At the level of individual attributes, the process of biotechnology entrepreneurship utilises two models, the individual model and the systemic model. The individual model is based on the psychological and nonpsychological attributes of the individual and the acquisition of entrepreneurial and commercialisation skills. The systemic model is based on the psychological and non-psychological attributes of the individual, in addition to a system of biotechnology entrepreneurship that provides entrepreneurial capabilities, commercialisation capabilities and support structures. Both models are driven by the environmental conditions that exist in the context of the study. Hence, while the individual model is practised in South Africa, the systemic model is practised in Brazil and the developed economies. At the level of entrepreneurial opportunities, the bioentrepreneurial opportunities are known in advance and differ in different contexts. Hence, while problem opportunities predominate in South Africa, efficiency opportunities predominate in Brazil. The entrepreneurial opportunities in the organising framework are not specific to the context and their discovery is dependent on the enterprising individual. At the level of environment for entrepreneurship, the themes related to the overall environment for biotechnology entrepreneurship, regulatory environment, funding, skills, and size of local market show that there are differences in the environment of entrepreneurship in the organising framework and the experience of the respondents in this research. While, the environment in the organising framework includes industry and the macro-environment, the environment for biotechnology entrepreneurship is broader and includes innovation and entrepreneurial conditions, in addition to industry and the macro-environment.

At the level of discovery, opportunity discovery in general entrepreneurship is dependent on the enterprising individual. However, this process in biotechnology entrepreneurship is dependent on R and D. R and D, in turn, is dependent on skills, infrastructure and funding, all of which are part of the innovation and entrepreneurial conditions. At the level of opportunity exploitation, this stage in general entrepreneurship is



Figure 2. A proposed theoretical framework of biotechnology entrepreneurship in South Africa and Brazil (adapted from Shane, 2003:11)

dependent on the individual entrepreneur. In contrast, the theme from this research shows that this stage in biotechnology entrepreneurship in South Africa and Brazil is dependent on effective collaboration among the key stakeholders. The existing literature (Müller et al., 2004; Ahn and Meeks, 2007; Ahn et al., 2010a; Afonso et al., 2010; Marcovich and Shinn, 2011; Afonso et al., 2012; Leydesdorff, 2012) on the exploitation of bioentrepreneurial opportunties shows that this requires a strategic alliance of government, research institutions, venture capitalists and large biotechnology companies. At the level of execution, the individual entrepreneur engages in resource assembly, organisational design and strategy (Shane, 2003) to create commercial value. In the process for biotechnology entrepreneurship, the collaboration among key stakeholders enables the efficient commercialisation of research through firm formation or licensing.

Furthermore, while the outcome of execution in general entrepreneurship satisifies the value creation need of the individual entrepreneur, the outcome of commercialisation of research in biotechnology entrepreneurship satisfies the need of multiple stakeholders in the form of economic development, financial return, and commercialisation (Ahn and Meeks, 2007), as well as social benefit. Gaps exist in current empirical literature and industry research related to the process of biotechnology entrepreneurship in South Africa and Brazil. In addition, there are differences between the process of general entrepreneurship, based on the organising framework of individual-opportunity nexus, and the process of biotechnology entrepreneurship in South Africa, based on the cross-case themes for this research. To address these gaps and differences, a theoretical framework of biotechnology entrepreneurship in South Africa and Brazil is proposed as shown in Figure 2. The proposed theoretical framework of biotechnology entrepreneurship in South Africa and Brazil (Figure 2) captures the process of biotechnology entrepreneurship in the developing economies of South Africa and Brazil. The individual and systemic models of biotechnology entrepreneurship are driven by environmental conditions. Given that the systemic model is operational in developed economies, it may mean that favourable environmental conditions support the emergence of the systemic model, while

unfavourable environmental conditions support the emergence of the individual model.

The bioentrepreneurial opportunities are known in advance and are often informed by the needs and priorities of the country, which are both social and economic. The approach in choosing which bioentrepreneurial opportunities to focus on should be informed by the areas that would make the biggest impact on the problems of the country. Another solvina consideration would be areas of competitive advantage such as biodiversity in South Africa and Brazil. The environment is where most of the differences occur between different contexts. This is because most of the factors are environmental factors and the extent to which the environment is conducive to bioentrepreneurial development determines to a large extent the ability to attract foreign skills, other sources of funding and large biotechnological companies and the ability to develop entrepreneurial and commercialisation capabilities within the system, and ultimately the success of biotechnology entrepreneurship. The government often plays a big role in determining the effectiveness of the environmental factors. The individual attributes and the environment determine the effectiveness of R and D in the next step. R and D is dependent on skills, infrastructure and funding. This is the defining point of biotechnology entrepreneurship, as the timeframe, outcome and cost can all be indeterminate, with no guaranteed outcome. Most of the costs incurred in biotechnology entrepreneurship occur at this stage; hence, multiple funding sources are often required to drive R and D. R and D spend and intensity are often used as measures to determine how committed countries are to researchintensive industries such as biotechnology, and often determine the output.

In South Africa and Brazil, the abundance of genetic materials as raw materials for R and D, through the biodiversity, is considered a competitive advantage. Only successful outcomes at the R and D stage lead to the exploitation of bioentrepreneurial opportunities. Although, alliances needed to the strategic exploit the bioentrepreneurial opportunities is the next step, the R and D step often requires funding from multiple stakeholders who will eventually participate in the exploitation of the successful output of the R and D. The exploitation of bioentrepreneurial opportunities requires effective collaboration among the key stakeholders, specifically a strategic alliance of government, research institutions, venture capitalists and large companies. At this stage, considerable resources are needed to go from the laboratory to the market and this works better in a systemic model than in an individual model, as the capabilities required are often beyond an individual. The availability of all these stakeholders defines the process of biotechnology entrepreneurship in the developed economies. In this study, Brazil has an availability of stakeholders while South Africa does not. That may

explain the different approaches in these two developing economies in terms of adopting the individual or systemic approach. The stage of bioentrepreneurial exploitation leads to commercialisation of research, which is the final stage in the model. The success of the stage of commercialisation of research is dependent on the success of the preceding stages in the framework. This is achieved either through firm formation or licensing. The different stakeholders have different needs, which are realised through successful commercialisation of the research.

The government is involved in most of the stages of the framework in South Africa and Brazil, through multiple such as facilitator, funder and buyer of roles biotechnology products. The extent of government involvement is determined by the specific context of each country. The triple helix of university, industry, government relations plays a key part in biotechnology entrepreneurship because of the high level of collaboration required. In this study, the triple helix of university, industry, government relations is controlled by the government, which creates a hybrid of triple helix I and III in Brazil while South Africa implements a triple helix I model. The direction of the arrows in Figure 2 shows that the individual attributes, the bioentrepreneurial opportunities and the environment are deemed to affect all the stages of biotechnology entrepreneurship entrepreneurial exploitation from R&D to and commercialisation of research. While, this does not prove causality, the importance of the individual attributes, entrepreneurial opportunities and the environment to entrepreneurship is supported by previous studies (Liebeskind et al., 1996; Audretsch and Stephan, 1998; Agrawal, 2001; Shane, 2003; Müller et al., 2004; Rothaermel and Deeds, 2004; Powell et al., Ahn and Meeks, 2007; Sytch and Bubenzer, 2008).

The management strategies adopted by the bioentrepreneurs in response to the environmental variables in South Africa and Brazil

The environmental variables in South Africa and Brazil resulted in different management strategies adopted by the bioentrepreneurs as summarised in Tables 4 and 5. The business management strategies adopted by bioentrepreneurs in emerging market biotechnology entrepreneurship can be broadly categorised into a system approach and an individual approach to business management, as articulated under theme 1.

CONCLUSION AND IMPLICATIONS

The methodological approach for this research was designed to achieve an in-depth understanding of the process of biotechnology entrepreneurship in South

No.	Environmental variable	Strategy adopted by bioentrepreneurs
i.	Lack of direction and leadership from the government	Most of the respondents avoid interaction with the government agencies at all cost. This impacts the university, industry, government relationships negatively
		Bootstrap funding is prevalent and some of the respondents seek funding abroad through donor agencies, venture capitalists and equity investors.
ii.	Lack of appropriate funding	overall cost and shorter timeframe such as diagnostics.
		Most of the bioentrepreneurs try to get to a later stage of development before seeking government funding as this gives them a better chance of securing funding from the government
iii.	Lack of skills	The bioentrepreneurs in South Africa tend to manage all aspects of the business from the core research responsibilities to the entrepreneurial and commercialisation responsibilities, and auxiliary services
iv.	Lack of a developed market for biotechnology products and solution	Most bioentrepreneurs compete in the international market with particular focus on Africa and other developing economies due to their focus on mostly problem opportunities
v.	Lack of infrastructure and support structures	Bioentrepreneurs employ similar strategy to iii above
vi.	An inclination by the universities to prioritise research publication over commercialisation	This leads to most of the bioentrepreneurs operating outside of the university as private entities which further strains the triple helix of university, industry, government relationship
vii.	The loss of skills through the brain drain	Bioentrepreneurs employ similar strategy to iii above
viii.	Low levels of commercialisation of research	The unfavourable environmental variables leads to limited capacity by the bioentrepreneurs to increase the commercialisation activities
ix.	Unfavourable policy and regulatory environment	Bioentrepreneurs employ similar strategy to i above
х.	Poor implementation of policies	Bioentrepreneurs employ similar strategy to i above
xi.	Lack of some of the stakeholders involved in biotechnology entrepreneurship	Bioentrepreneurs employ similar strategy to iii above
xii.	Low levels of R&D spend	Bioentrepreneurs employ similar strategy to viii above

Table 4. The strategies adopted by the bioentrepreneurs in response to the environmental variables in South Africa.

Africa and Brazil in their original contexts. The empirical, methodological and theoretical contributions, as well as the implications for the different stakeholders are articulated below.

Empirical contributions to the literature

The main empirical contribution of this research to the literature is the contribution to the body of knowledge, which addresses the gap created by the paucity of empirical research on biotechnology entrepreneurship in the context of developing economies. Few empirical research studies are specific to biotechnology entrepreneurship (Schoemaker and Schoemaker, 1998; Müller et al., 2004; Audretsch et al., 2008; Carsrud et al., 2008; Oliver, 2008; Gunn et al., 2013). Furthermore, the empirical research that is specific to biotechnology

entrepreneurship, most studies are specific to the developed economies and few are specific to the developing economies (Onyeka, 2011). The paucity of empirical research on biotechnology entrepreneurship in the developing economies' context creates a gap that this research addressed.

Methodological contributions to the literature

The few studies on biotechnology entrepreneurship, in developing economies, employ mostly survey methodology and single case studies within one developing economy. The use of qualitative multiple case studies, in the idiographic philosophical tradition, in two developing economies, is deemed to enrich the discourse in biotechnology entrepreneurship and hence make a contribution to the knowledge of biotechnology Table 5. The strategies adopted by the bioentrepreneurs in response to the environmental variables in Brazil.

No.	Environmental variable	Strategy adopted by bioentrepreneurs
i.	Effective direction and leadership from the government	Most of the respondents in Brazil work very closely with the government agencies and often work for the government at the same time as they are working for the university or the industry. This has a very positive impact on the university, industry, government relationships and has resulted in the successful completion of many internationally-acclaimed collaborative projects
ii.	Availability of appropriate funding from different sources	Bioentrepreneurs in Brazil are not constrained by a lack of appropriate funding and hence concentrate efforts on getting to results that help solve the national problems and areas of focus, given that government plays a big role in providing appropriate funding
iii.	Lack of aggregate skills	There is a high level of local and international collaboration to alleviate the effect of aggregate skills shortage. The bioentrepreneurs in Brazil also help drive skills exchange programmes that helps build sustainable capacity for the future
iv.	Availability of developed market for biotechnology products and solution	The bioentrepreneurs in Brazil focus primarily on the areas of biotechnology entrepreneurship prioritised by the government in the biotechnology development policy of Brazil. This creates an alignment that caters for national imperatives as well as make Brazil more competitive in the international market
v.	Availability of infrastructure and support structures	Bioentrepreneurs concentrate efforts on getting to results that help solve the national problems and areas of focus; as well as position the country competitively in the international market
vi.	A national culture that considers entrepreneurship to be impure	Bioentrepreneurs tend to work at the university, government and the industry at the same time in order not be seen as being after getting rich
vii.	High cost of funding	Bioentrepreneurs are very cautious about setting up their own biotechnology companies. The tendency is to allow the big biotechnology companies to cushion the funding cost in a collaborative set-up
viii.	Bureaucratic and inefficient processes by government agencies	Bioentrepreneurs in Brazil seek services like patent registration abroad due to the length of time it takes to do this in Brazil
ix.	Unfavourable policy and regulatory environment	The bioentrepreneurs use the good triple helix relationships to effect changes in the policy and regulatory environment of biotechnology entrepreneurship in Brazil
x.	Availability of all the stakeholders involved in biotechnology entrepreneurship	Bioentrepreneurs in Brazil utilise the contributions of the different stakeholders to optimise the outcome of collaborative projects
xi.	High level of R&D funding relative to other emerging economies; but low in comparison to OECD countries	Bioentrepreneurs in Brazil employ similar strategies to ii, iv and v above

entrepreneurship in developing economies.

A qualitative multiple case study method was used for this study, at a country level of analysis. The use of a holistic multiple case study approach provided the opportunity for literal and theoretical replication (Yin, 2009). The case selection was purposefully aimed at good candidates for biotechnology entrepreneurship in the developing economies with sufficient similarity, and variability, to provide a suitable context for this research (Yin, 2009).

Theoretical contributions to the literature

The main theoretical contribution of this research is the development of a theoretical framework of biotechnology entrepreneurship, which defines the dynamics of

biotechnology entrepreneurship in South Africa and Brazil (Figure 2). The research followed the idiographic tradition, defining themes that are tested against literature. The proposed theoretical framework of biotechnology entrepreneurship is based on the themes that emerged from the cross-case analysis of the process of biotechnology entrepreneurship in South Africa and The within-case analysis of each case Brazil. incorporated the lived experiences of the bioscientists, bioentrepreneurs, and subject matter experts SMEs in biotechnology entrepreneurship in South Africa and Brazil. At the time of this research, there is no known theoretical framework of biotechnology entrepreneurship, especially from a developing economies' context. Although this theoretical framework cannot be generalised to all developing economies, it provides the means to study the process of biotechnology entrepreneurship in

other developing and developed economies.

Policy and other implications for the government

One of the themes identified through the cross-case analysis is "the government plays an important role in biotechnology entrepreneurship in South Africa and Brazil". In addition, the policy and regulatory environment was among the top challenges identified for the development of biotechnology entrepreneurship in the developing economies.

The role of government in providing a favourable environment for the development of biotechnology entrepreneurship includes legislation on the national biotechnology strategy, policies on public research institutions, policies related to research funding, intellectual property policies, regulations on universityindustry technology transfer, regulations on taxation, labour laws, policies on the acquisition of scarce skills, policies on science and mathematics in the education curriculum, and policies related to regional and international collaboration on biotechnology entrepreneurship.

The availability of the relevant policies and regulations; the effectiveness of implementing these policies by the government agencies; and the leadership and direction provided by government were highlighted as some of the differences between the developed and developing economies.

Furthermore, the government plays the roles of facilitator, buyer of biotechnology products and solutions, and funder, in addition to the role of providing a favourable policy and regulatory environment. The facilitation role of the government includes the provision of leadership, infrastructure. capacity. research institutions and platforms for local and international collaborations among key stakeholders. The role of government as a buyer is necessitated by the social obligation of government to provide improved healthcare, food security, energy sufficiency and sustainable environmental practices. The role of the government as a funder includes the funding of R and D, infrastructure funding and project funding. It is recommended that the government of South Africa, in particular, find effective ways of delivering on these roles to create a conducive environment for the development of biotechnology entrepreneurship.

Implications for the other stakeholders

The empirical contribution of this research has implications for the other stakeholders involved in the biotechnology industry, such as the research institutions, venture capitalists, large biotechnology companies and bioentrepreneurs. A clearer understanding of the dynamics of biotechnology entrepreneurship in developing

economies is expected to aid decision making related to the biotechnology entrepreneurship by these stakeholders. There are multiple points of stakeholder collaboration and strategic alliances in the proposed theoretical framework for biotechnology entrepreneurship. These are in the environment for biotechnology entrepreneurship; R and D; opportunity exploitation; and commercialisation of research. The research institutions need to review their policies on intellectual property and the transfer of technology from the university to the industry, in alignment with similar policies by the government. This was highlighted as one of the gaps in the development of biotechnology entrepreneurship in these developing economies. Furthermore, the culture of the universities and research institutions needs to be changed from prioritising publication to being focused on commercialisation of research to realise economic and social value. The understanding of the dynamics of biotechnology entrepreneurship in these developing economies, especially in South Africa where there is a lack of a developed venture capital industry, will aid the venture capitalists in understanding the peculiarities of the environment, the challenges and gaps, the role of the government and the opportunities that can be exploited. Importantly, the lessons from the success of the venture capital market in Brazil can be implemented in South Africa, given an enabling regulatory environment.

The absence of large biotechnology companies was highlighted as one of the gaps in South Africa. Given the involvement of the large biotechnology companies in major R and D; their role as cooperation partners, customer and competitor; and the availability of good research universities and skilled researchers in South Africa, there is an opportunity for the large biotechnology companies to seek out collaboration opportunities in South Africa, provided the enabling environment expected to be provided by the government is in place. Similarly, they can use their expertise in Brazil to enter the South African biotechnology industry.

Conflict of Interests

The author(s) have not declared any conflict of interests.

REFERENCES

- Ács ZJ, Audretsch DB (2003). Handbook of entrepreneurship research: an interdisciplinary survey and introduction, Kluwer Academic Publishers, USA.
- Afonso O, Monteiro S, Thompson M (2010). A Growth Model for the Quadruple Helix Innovation Theory, Working Papers (FEP) ---Universidade do Porto (271): 1-21.
- Afonso O, Monteiro S, Thompson M (2012). A growth model for the quadruple helix. J. Bus. Econ Mngmt. 13(5): 849-865.
- Agrawal AK (2001). University to industry knowledge transfer: literature review and unanswered questions. Intl. J. Mngmt. Rev. 3(4): 285-302.
- Ahn M, York A (2011). Resource-based and institution-based approaches to biotechnology industry development in Malaysia. Asia.

Pacific. J. Mngmt. 28(2): 257-275.

- Ahn MJ, Hajela A, Akbar M (2012). High technology in emerging markets. Asia. Pacific. J. Bus. Admin. 4(1): 23-41.
- Ahn MJ, Meeks M (2007). Building a conducive environment for life science-based entrepreneurship and industry clusters. J. Comm. Biotechnol. 14(1): 20-30.
- Ahn MJ, Meeks M, Bednarek R, Ross C, Dalziel S (2010a). Towards a high-performance bioeconomy. Intl. J. Comm Mngmt. 20(4): 308-330.
- Ahn MJ, Meeks M, Davenport S, Bednarek R (2010b). Exploring technology agglomeration patterns for multinational pharmaceutical and biotechnology firms. J. Comm. Biotechnol. 16(1): 17-32.
- Audretsch DB, Stephan PE (1998). How and why does knowledge spill over? The case of biotechnology. CEPR Discussion Papers.
- Audretsch DB, Taylor Aldridge T, Perry M (2008). A survey review of university biotechnology and entrepreneurship commercialization. Handbook, Bioentrepreneurship. pp.175-187
- Barro RJ, Lee JW (2000). International data on educational attainment updates and implications. National Bureau of Economic Research Cambridge, Mass., USA.
- Battelle/Biotechnology Industry Organisation (2010). State Bioscience Initiatives. http://www.bio.org
- Battelle/Biotechnology Industry Organisation (2012). State Bioscience Industry Development. http://www.bio.org
- Biotechnology Industry Organisation (2008). Guide to biotechnology. http://www.bio.org
- Bosma N, Levie J (2010). Global Entrepreneurship Monitor 2009: executive report. Global Entrepreneurship Research Association (GERA).
- Bud R (1991). Biotechnology in the twentieth century. Soc. Stud. Sci. 21(3): 415 457.
- Carree MA, Thurik AR (1998). Small firms and economic growth in Europe. Atlantic Economic. J. 26(2): 137-146.
- Carsrud AL, Brännback M, Renko M (2008). Strategy and strategic thinking in biotechnology entrepreneurship. Handbook of bioentrepreneurship. pp.81-101
- Clarke BR (2002). Transfer to Africa of the resources and rewards from biotechnology: the need for a participatory approach. J. Comm. Biotechnol. 9(1): 31-39.
- Creswell JW (2009). Research design: qualitative, quantitative, and mixed method approaches. Third ed., Sage Publications, Inc. Thousand Oaks, CA.
- Cunningham JB, Lischeron J (1991). Defining entrepreneurship. J. Small. Bus. Mngmt. 29(1): 45 61.
- Department of Science and Technology (2001). National Biotechnology Strategy for South Africa. http://www.dst.gov.za/publicationspolicies/strategies-reports/dst_biotechnology_strategy.pdf
- Dibner MD (1986). Biotechnology in Europe. Science. 232(4756): 1367 1372.
- Dunham L, Ahn M, York AS (2012). Building a bioeconomy in the heartland. J. Enterpris.Communit.. 6(1): 84-100.
- Eckhardt JT, Shane SA (2003). Opportunities and entrepreneurship. J. Mngmt. 29(3): 333 349.
- Eisenhardt KM (1989). Building theories from case study research. The Acad. Mngmt. Rev. 14(4): 532-550.
- Eisenhardt KM (1991). Better stories and better constructs: the case for rigor and comparative logic. The Acad. Mngmt. Rev. 16(3): 620-627.
- Eisenhardt KM, Graebner ME (2007). Theory Building from Cases: Opportunities and Challenges. Acad. Mngmt. J. 50(1): 25-32.
- Ernst & Young (2006). South African Biotech Review, Discussions with Industry Stakeholders. http://www.biopad.org.za/documents/Ernst_&_Young.pdf
- Ernst & Young (2010a). Global Review: Entrepreneurship. http://www.ey.com
- Ernst & Young (2010b). Beyond Borders: Global Biotechnology Report. http://www.ey.com
- Fontes M (2001). Biotechnology entrepreneurs and technology transfer in an intermediate economy. Technological forecasting and social change. 66: 59-74.
- Gastrow M (2008). Great expectations: the state of biotechnology research and development in South Africa. Afr. J. Biotechnol. 7(4): 342-348.
- Gunn MA, Dever J, Tzagarakis-Foster C, Lorton Jr P, Kane K,

Masterson N (2013). An agile, cross-discipline model for developing bio-enterprise professionals. J. Comm. Biotechnol. 19(4): 72-87.

- Hannan MT, Freeman J (1987). The ecology of organizational founding: American labor unions, 1836-1985. Am. J. Sociol. 92(4): 910-943.
- Herrington M, Kew J, Kew P (2008). Global Entrepreneurship Monitor: 2008 South African report. Graduate School of Business, University of Cape Town, Cape Town.
- Hwang S (2008). Utilizing Qualitative Data Analysis Software A Review of Atlas. ti. Soc. Sci. Comp. Rev. 26(4): 519-527.
- Kelley DJ, Singer S, Herrington MD (2012). The Global Entrepreneurship Monitor. 2011 Global Report.
- Kenney M (1986). Schumpeterian innovation and entrepreneurs in capitalism. A case study of the U.S. biotechnology industry. Res.Policy. 15: 21-31.
- Kettler H, Casper S (2001). Turning good science into successful businesses: the technology transfer systems in the UK and Germany. J. Comm. Biotechnol. 7(3): 197-207.
- Kivinen O, Varelius J (2003). The emerging field of biotechnology the case of Finland. Science, Technology & Human Values. 28(1): 141 -161.
- Klonoski R (2013). The Case For Case Studies: Deriving Theory From Evidence. J. Bus. Case. Studies (Online). 9(3): 261-n/a.
- Leydesdorff L (2012). The Triple Helix, Quadruple Helix, ..., and an N-Tuple of Helices: explanatory models for analyzing the knowledgebased economy? J. Knowl. Econ. 3(1): 25-35.
- Liebeskind JP, Oliver AL, Zucker L, Brewer M (1996). Social networks, learning, and flexibility: sourcing scientific knowledge in new biotechnology firms. Organization Science. 7(4): 428-443.
- Marcovich A, Shinn T (2011). From the Triple Helix to a Quadruple Helix? The case of Dip-Pen Nanolithography. Minerva: A Review of Science, Learning & Policy. 49(2): 175-190.
- Meyers A (2012). The birth of a discipline: Bioentrepreneurship. J. Comm. Biotechnol. 18(4): 3-4.
- Müller C (2002). The evolution of the biotechnology industry in Germany. TRENDS in Biotechnol. 20(7): 287-290.
- Muller C, Fujiwara T (2002). The commercialization of biotechnology in Japan. Drug Discovery Today. 7(13): 699-704.
- Müller C, Fujiwara T, Herstatt C (2004). Sources of bioentrepreneurship: the cases of Germany and Japan. J. Small. Bus. Mngmt. 42(1): 93-101.
- Nilsson A (2001). Biotechnology firms in Sweden. Small Busin. Econ.. 17(1): 93-103.
- Nilsson AS, Rickne A, Bengtsson L (2010). Transfer of academic research: uncovering the grey zone. J. Tech. Trans. 35(6): 617-636.
- Oliver AL (2008). University-based biotechnology spin-offs. Handbook of bioentrepreneurship. pp.188-205
- Onyeka CJ (2011). Biotechnology commercialisation in universities of developing countries: A review of the University of Ibadan, Nigeria. J. Comm. Biotechnol. 17(4): 293-300.
- Organisation for Economic Cooperation and Development (2009). Biotechnology Statistics 2006. http://www.oecd.org/dataoecd/4/23/42833898.pdf
- Organisation for Economic Cooperation and Development (2013a). Biotechnology R&D Expenditures in the Business Sector. http://www.oecd.org/sti/biotech/keybiotechnologyindicators.htm
- Organisation for Economic Cooperation and Development (2013b). Share of countries in biotechnology patents filed under PCT, 2009-11. http://www.oecd.org/sti/biotech/keybiotechnologyindicators.htm
- Phan PHC, Venkataraman S, Velamuri SR (2008). Entrepreneurship in emerging regions around the world: theory, evidence and implications. Edward Elgar Publishers, USA.
- Pisano GP (1990). The R&D boundaries of the firm: an empirical analysis, Administrat. Sci. Quarterly. 35(1): 153 176.
- Pisano GP (1991). The governance of innovation: vertical integration and collaborative arrangements in the biotechnology industry. Res. Policy. 20(3): 237-249.
- Powell WW, White DR, Koput KW, Owen-Smith J (2005). Network dynamics and field evolution: the growth of interorganizational collaboration in the life sciences. Amer. J. Sociol. 110(4): 1132-1205.
- Rambaree K (2007). Bringing Rigour in qualitative social research: the use of a CAQDAS. UOM. Res. J. 13.
- Rothaermel FT, Deeds DL (2004). Exploration and exploitation alliances

in biotechnology: a system of new product development. Strat. Mngmt. J. 25(3): 201-221.

- Schoemaker HJP, Schoemaker AF (1998). The three pillars of bioentrepreneurship. Nature biotechnology. 16: 13-15.
- Shane S, Venkataraman S (2000). The promise of enterpreneurship as a field of research. The Acad. Mngmt. Rev. 25(1): 217-226.
- Shane SA (2003). A general theory of entrepreneurship: the individualopportunity nexus. Edward Elgar Publishers, UK.
- Stake RE (2006). Multiple case study analysis. The Guilford Press, New York.
- Sytch M, Bubenzer P (2008). Research on strategic alliances in biotechnology: an assessment and review. Handbook, Bioentrepreneurship. pp.102-128
- Tushman ML, Anderson P (1986). Technological discontinuities and organizational environments. Administrat. Sci. Quarterly. 31(3): 439-465.
- Weitzman EA (1999). Analyzing qualitative data with computer software. Health Services Research. 34(5p2): 1241.
- Welch C, Piekkari R, Plakoyiannaki E, Paavilainen-mäntymäki E (2011). Theorising from case studies: Towards a pluralist future for international business research. J. Intl. Bus. Stu. 42(5): 740-762.

- Yin RK (2009). Case study research: design and methods, Applied Social Research Methods Series, Vol 5, Fourth ed. Sage Publications, Inc. Thousand Oaks, CA.
- Zdenek K (2008). Making Thinking Visible with Atlas.ti: Computer Assisted Qualitative Analysis as Textual Practices. Forum : Qualitat. Social Res.9(2).
- Zylberberg E, Żylberberg C, Oner AC (2012). Biotechnology in Brazil: An industry overview. J. Comm. Biotechnol. 18(4): 9-18.

African Journal of Business Management

Related Journals Published by Academic Journals

Journal of Geography and Regional Planning
Journal of Economics and International Finance
Journal of Hospitality Management and Tourism
International Journal of Sociology and Anthropology
Journal of Public Administration and Policy Research
African Journal of Marketing Management